Final Report

Global Challenges Research Fund Evaluation

Module: Gender Equality, Social Inclusion and Poverty Audit

Date: 16 February 2021 Authors: Mary Ann Brocklesby, David Walker, Douglas Elsey, Ekaterina Shaleva, Violeta Fernandez

Submitted by Itad In association with







African Institute for Development Policy Bridging Development Research, Policy and Practice







Acknowledgements

The GESIP module team would like to thank all key informants at delivery partner and award level for their active and helpful participation in interviews and co-creation workshops, as well their support in providing relevant documentation. In particular, we would also like to thank the award holders who joined us for the post-finding discussion workshop.

Disclaimer

The views expressed in this report are those of the evaluators. They do not represent those of Itad or of any of the individuals and organisations referred to in the report.

'Itad' and the tri-colour triangles icon are a registered trademark of ITAD Limited.

Contents

	List of	acronyms	iii
Ex	ecutive Su	nmary	5
1	Introducti	on	9
	1.1	Strategic and policy context for the GCRF GESIP audit in 2020–21	10
	1.2	Addressing GESIP within the GCRF: key principles and findings of previous GESIP assessments	12
	1.3	Structure of the report	15
2	Methodol	ogy	16
	2.1	Overview of the GESIP audit approach	16
	2.2	Overview of data collection	21
	2.3	Approach to data analysis	23
	2.4	Strengths and limitations of the approach	23
3	Results		25
-	3.1	EQ 1: How are gender mainstreaming activities currently being strengthened throughout the fund?	25
	3.2	EQ 2: How Effectively is Poverty Framed and Understood within the GCRF ecosystem	34
	3.3	EQ 3: How is social inclusion/equality, diversity and inclusion framed and understood within the GCRF ecosystem?	38
	3.4	To what extent is GCRF responsive to GESIP issues with respect to Covid-19 adaptation	ıs?45
	3.5	Interlinkages with other modules	45
4	GESIP aud	lit conclusions	47
	4.1	Gender equality: conclusion, improvement opportunities, and ongoing gaps and challe 48	nges
	4.2	Poverty alleviation: conclusion, improvement opportunities, and ongoing gaps and challenges	50
	4.3	Social inclusion/equality, diversity and inclusion: conclusion, improvement opportunities, and ongoing gaps and challenges	51
	4.4	GESIP-sensitive Covid-19 responses: conclusion, improvement opportunities, and ongoing gaps and challenges	53
5	Recomme	ndations	54
	5.1	Overview	54
	5.2	Recommendations	54
Re	eferences		58

List of acronyms and abbreviations

AHRC	Arts and Humanities Research Centre
AMS	Academy of Medical Sciences
BA	British Academy
BBSRC	Biotechnology and Biological Sciences Research Council
BEIS	Business, Energy and Industrial Strategy (formerly BIS)
CI	Co-investigator
Covid-19	Coronavirus Disease 2019
DAC	Development Assistance Committee
DEFNI	Department for the Economy Northern Ireland
DFID	Department for International Development
DP	Delivery Partner
EDI	Equality, Diversity and Inclusion
EPSRC	Engineering and Physical Sciences Research Council
EQ	Evaluation Question
ESRC	Economic and Social Research Council
FCDO	Foreign, Commonwealth & Development Office
FCO	Foreign & Commonwealth Office
FLAIR	Future Leaders African Independent Research
GAM	Gender Analysis Marker
GCRF	Global Challenges Research Fund
GDP	Gross Domestic Product
GE	Gender Equality
GEDI	Gender Equality, Diversity and Inclusion
GESIP	Gender Equality, Social Inclusion and Poverty (Alleviation)
GtR	Gateway to Research
HEFCW	Higher Education Funding Council for Wales
HEPSSA	Higher Education Partnerships in Sub-Saharan Africa
IATI	International Aid Transparency Initiative
ICAI	Independent Commission for Aid Impact
IPP	International Partnership Programme
KII	Key Informant Interview
KPI	Key Performance Indicator

LMIC	Low and Middle-Income Country
M&E	Monitoring and Evaluation
MEL	Monitoring, Evaluation and Learning
MEQ	Main Evaluation Question
MRC	Medical Research Council
NERC	Natural Environment Research Council
ODA	Official Development Assistance
OECD	Organisation for Economic Co-operation and Development
PI	Principal Investigator
R&I	Research and Innovation
RAEng	Royal Academy of Engineering
RODA	Reporting Official Development Assistance
RS	Royal Society
SDG	Sustainable Development Goal
SFC	Scottish Funding Council
SI	Social Inclusion
STFC	Science and Technology Facilities Council
ТоС	Theory of Change
ToR	Terms of Reference
UK	United Kingdom
UKRI	UK Research and Innovation
UKSA	UK Space Agency
VfM	Value for Money

Executive summary

The approach to the combined gender, social inclusion and poverty (GESIP) module of the Global Challenges Research Fund (GCRF) evaluation is a 'social audit'. This audit assesses the extent to which poverty, inequality and other forms of social exclusion (including disability) are addressed in the planned outcomes and the pathways to impact of (GCRF) programmes and investments.

The audit accompanies three other modules (each of which is focused on the topics of management, relevance and fairness) in seeking to review the £1.5 billion United Kingdom (UK) government fund – established in late 2015 to support cutting-edge research that addresses the challenges faced by developing countries. The GCRF forms part of the UK's official development assistance (ODA) commitment and contributes to the achievement of the UK's 2015 aid strategy's goals.

The GCRF ensures that UK science takes a leading role in addressing the challenges faced by developing countries while also developing the UK's ability to deliver cutting-edge research and innovation (R&I) for sustainable development. The GCRF is implemented by 17 of the UK's R&I funders, which commission R&I as delivery partners (DPs).

Background

The national policy discourse on ODA has been evolving since GCRF started in 2015. Several significant changes have taken place since 2020, with the implications for GCRF still emerging. First, in February 2020 the UK government announced the Integrated Review of foreign policy, defence, security and international development. Second, the merger in August 2020 of the Foreign & Commonwealth Office (FCO) and Department for International Development (DFID) into the new Foreign, Commonwealth and Development Office (FCDO) is expected to be central to the delivery of this emerging vision.

In parallel, a number of reviews have provided insights from which to explore the broader aspects of GESIP across the GESIP ecosystem. The 'Foundation Evaluation of GCRF', the 'BEIS Gender Equality Review of GCRF and Newton Fund' and external evidence reviews conducted by UK Research and Innovation (UKRI) on the key challenges for equality, diversity and inclusion in the UK and internationally¹ collectively show that the GCRF's policies and systems did not directly address GESIP issues at the inception of the fund, although all three dimensions are implicit in its objectives and goals. This audit continues and builds on the trajectory of these previous reviews.

¹ Itad (2019), Tetra Tech (2020), Guyan & Oloyede (2019) and Moody & Aldercotte (2019) respectively.

Methodology

The GESIP audit was designed to provide an audit of the organisational layers of GCRF, with a particular focus on a sample of DPs and award holders. The audit assesses how: strategies, policies and processes have provided a consistent strategic focus for GESIP; what expertise is available to support award holders integrate GESIP issues into their research; how far monitoring, evaluation and learning (MEL) activities are supporting improvements in capturing the GESIP dimensions of research outcomes; and the extent to which the GESIP dimensions of research are being communicated and disseminated across GCRF and externally.

The GESIP Audit addresses four sub-evaluation questions (EQs) – each of which also included considerations for adapting to Covid-19.

- EQ 1. How is gender equality mainstreaming currently being strengthened throughout the fund?
- EQ 2. How effectively is poverty framed and understood within the GCRF ecosystem?
- EQ 3. How effectively is social inclusion framed and understood within the GCRF ecosystem?
- EQ 4. Recommendations: How can the targeting (poverty), gender sensitivity and social inclusion of GCRF be improved?

The audit draws upon four bodies of evidence: a DP literature review (313 documents from 16 DPs), an assessment of award topics (1143 awards reviewed using data science techniques), an award-level audit (201 documents across 54 awards, and engaging 45 female and 49 male award holders), and two co-creation workshops (a dozen participants).

Findings and conclusions

The conclusions below demonstrate that the GCRF ecosystem, as it currently stands, is moving positively towards an improved compliance culture in relation to ODA expectations on GESIP. Shifts in this direction are more evident with respect to gender equality concerns, and to a lesser extent across social inclusion/Equality, Diversity and Inclusion (SI/EDI) concerns. Compliance concerns regarding the framing and addressing of 'poverty' are shown to be the least fulfilled and fairly problematic.

The conclusions also show that some of the recommendations of the Tetra Tech 'Review of Approaches to Gender Equality' report (2020) and the EDI reviews (2019) have been gaining momentum.² For example, the positive trajectory of growth leading towards comprehensive systems, policies, expertise and MEL systems for GESIP-related compliance is demonstrating opportunities for improvement and consolidation (Tetra Tech recommendations 1–3). Gaining traction on the fourth recommendation of the Tetra Tech report ('dedicated senior management capacity, sufficient resourcing and clear accountabilities at all levels to design, implement, monitor and adapt the approach') will also be particularly challenging given challenges noted around the strategic development and maintenance of GESIP-related expertise, capacity building and monitoring systems.

Overall, although there are pockets of promising or best practice across all GESIP areas, there is still some way to go before GCRF can foreseeably demonstrate a shift from a culture of ODA

² Recommendation 1: BEIS should develop a clear, coherent approach to gender equality, diversity and inclusion at the ODA portfolio level.

Recommendation 2: BEIS should institute a systematic approach to learning, clearly differentiating between internal and external functions at key stages in the fund management cycle.

Recommendation 3: BEIS should develop a fund level approach to gender equality, diversity and inclusion that is tailored to each of the funds.

Recommendation 4: BEIS should ensure there is dedicated senior management capacity, sufficient resourcing and clear accountabilities at all levels to design, implement, monitor and adapt the approach.

compliance towards a culture of excellence in addressing GESIP concerns. The metaphor of a 'broken trail' is used frequently in the report to identify where specific attention is required. This term has variable meanings in the report, but 'broken trails' can refer to the unfulfilment of accountability measures being in place or acted upon (conclusions 1 and 3), the limited capture of quality MEL compliance data (conclusions 2 and 3), as well as gaps in compliance guidance between higher-level ODA commitments and award holders (conclusions 2 and 3).

Conclusion 1: Gender equality mainstreaming

The mainstreaming of gender equality concerns across GCRF is moving at a gradual pace and is being supported by a variety of mechanisms from the DP level through to award holders – although strategic steers at both levels are mostly uncoordinated. This lack of strategic guidance is leading to a 'broken trail' of MEL reporting and, thereby, accountability between the DP and award level. As a result, while progress is gradual, GCRF is broadly 'on track' at the DP level in terms of mainstreaming gender compliance concerns – but could move in a faster and more coordinated way while also celebrating 'excellence' as much as 'compliance' results with award holders.

Conclusion 2: Poverty

There is limited evidence of a coherent framing for poverty or 'poverty alleviation' guiding GCRF, its DPs and award holders. Under the umbrella of the Sustainable Development Goals (SDGs), poverty is implicitly and simplistically assumed to be core to all R&I activities undertaken through the GCRF, although there is a wide range of framing and understanding of poverty-related terminology across the system. As a result, there is continuing evidence of a 'broken trail' in the capture and dissemination of the poverty dimensions of GCRF-supported research alongside an under-reporting of poverty results and outcomes.

Conclusion 3: Social inclusion/equality, diversity and inclusion

The mainstreaming of SI/EDI compliance concerns is fragmented at both the DP and award holder level, although there are promising signs of downward diffusion of EDI language and concepts in a good number of DPs and awards. As a result, and in line with preceding conclusions, there is evidence of a 'broken trail' of managerial guidance and accountability with respect to the mainstreaming of SI/EDI issues across GCRF – particularly in relation to LGBTQIA+ concerns. Consequently, progress on the mainstreaming of SI/EDI compliance concerns is somewhat behind that of gender compliance mainstreaming concerns – particularly at the award level – while a small number of DPs and awards are actively endeavouring to innovate for 'excellence' in SI/EDI mainstreaming, rather than 'compliance'.

Conclusion 4: Adapting to Covid-19

Explicit guidance from DPs on adapting to a Covid-19 context with respect to the GESIP dimensions was rarely found or reported. The absence of guidance does not mean that dialogue and more formal guidance between some award holders and some DPs, as well as Challenge Leaders or Hubs, is non-existent: evidence from the award holder review suggests that GESIP-related dialogue was prominent, but largely driven horizontally by the interests of award holders (primarily in relation to methodological adaptations).

Recommendations

The recommendations are designed to support GCRF achieve and move from a culture of ODA compliance towards a culture of excellence in addressing GESIP concerns. They work to consolidate and strengthen the progress made to date to mainstream gender equality and address the social inclusion and poverty dimensions of ODA-supported research. They build on the Tetra Tech Review of Gender Equality Approaches in the GCRF and its guiding principles and Theory of Change (ToC). The recommendations are framed to inform the development of the GCRF's Gender,

Equality, Diversity and Inclusion (GEDI) Strategy, to optimise interlinkages between gender and social inclusion and poverty alleviation, and to address various 'broken trails' in the GCRF system.

Recommendation 1: Embed GESIP in the GCRF principles

The ongoing development of the GEDI strategy offers the opportunity to review the narrative of GCRF's key principles (Challenge and Impact Focus, Interdisciplinary Research Excellence, Global Partnerships, Organisation and Leadership) in relation to the objective of making explicit the GESIP-related compliance expectations of the fund. The process of reframing the key principles narrative should underpin and run in parallel with the development of the GEDI strategy. This will optimise cross-fertilisation between the two activities and build ownership across the whole GCRF ecosystem for the GEDI strategy.

Recommendation 2: Integrate a clear poverty perspective into the GEDI strategy

Poverty compliance guidance constitutes the weakest part of the GESIP compliance infrastructure across GCRF. By further clarifying GCRF's expectations and framing in relation to the interlinkages between poverty on the one hand and gender equality and SI/EDI on the other, a broader poverty impact narrative can be developed over time. This foundation provides an avenue for GCRF to assess both compliance and excellence. While a critical interface will be at the call and in the proposal stage, clarity is needed as to how these interlinkages apply at all stages of the funding management cycle.

Recommendation 3: Standardise the MEL infrastructure and communication of GESIP expectations (and associated resource support) across the GCRF and its funding cycles

Standardising these expectations throughout the GCRF system is fundamental to consolidating and expanding the emerging and promising practices among some DPs in the way GESIP expectations are communicated. It is recommended that GCRF strengthen and standardise activities at the pre-proposal stage, extend the engagement time with researchers on the approach and design of the proposal with respect to GESIP concerns, extend the use of gender champions to include SI/EDI and poverty champions, build and facilitate access to GESIP expertise and knowledge repositories, and embed a GESIP focus in the redesign of the reporting official development assistance (RODA) system. GCRF could also work with DPs to increase the accuracy, visibility and – where possible – the complexity of disability and LGBTQIA+ data categories in the Business Energy and Industry Strategy (BEIS) tracker and ResearchFish platform.

Recommendation 4: Open up learning spaces for GESIP that are facilitative and reflexive as part of an increased emphasis on GESIP-oriented knowledge management for both compliance and excellence

Building on observed and growing communities of practice and horizontal learning mechanisms, GCRF can with DPs and award holders to identify entry points and actions for developing a GESIP focus within GCRF's knowledge management system. This recommendation supports recommendation 2 of the Tetra Tech (2020) Approaches to Gender Equality Review – '*BEIS should institute a systematic approach to learning, clearly differentiating between internal and external functions at key stages in the fund management cycle*'. A more GESIP-focused knowledge management system will serve to better coordinate actions for capturing and disseminating GESIP-specific lessons on compliance and excellence.

1 Introduction

This is the Global Challenges Research Fund (GCRF) Gender Equality, Social Inclusion and Poverty (GESIP) audit report, one of four modules in the GCRF Evaluation, Stage 1a. The following section provides background on the strategic and policy context surrounding the GCRF with respect to GESIP issues in recent years, and provides a summary of findings from previous relevant assessments.

Stage 1a assesses GCRF's core commissioning and management functions – the activity level in the Theory of Change (ToC) – via four modules that focus on management, relevance and targeting, fairness and the integration of gender, social inclusion and poverty as core concerns in the fund. The main evaluation question (MEQ) for Stage 1a asks: **is GCRF relevant, well targeted, fair, gender sensitive, socially inclusive and well managed**?

The aim is to provide a learning (formative) assessment to ensure that the conditions are in place to support GCRF's outcomes and impact. Stage 1a will produce an in-depth view of how GCRF works as a fund, where it is working well and where it could be improved (Box 1 provides an overview of the GCRF Evaluation).

The GESIP audit and the other three modules together contribute to addressing Stage 1a's MEQ 1: 'Is the GCRF relevant, coherent, well targeted, fair, gender sensitive, socially inclusive and well managed?'

The aim of the GESIP audit is to provide a light touch assessment of the extent to which the GESIP concerns have been addressed throughout GCRF at all levels of operation between 2016 and 2020.

The GESIP audit addresses four sub-evaluation questions (EQs) under the MEQ as follows:

- EQ 1. How is gender equality mainstreaming currently being strengthened throughout the fund?
- EQ 2. How effectively is poverty framed and understood within the GCRF ecosystem?
- EQ 3. How effectively is social inclusion framed and understood within the GCRF ecosystem?

The fourth EQ – 'How can poverty, gender sensitivity and social inclusion of GCRF be improved' – is addressed in the conclusions and recommendations of this audit report.

Box 1. Overview of GCRF and the evaluation

GCRF is a £1.5 billion fund announced by the United Kingdom (UK) government in late 2015 to support cutting-edge research that addresses the challenges faced by developing countries. GCRF forms part of the UK's Official Development Assistance (ODA) commitment and contributes to the achievement of the UK's 2015 aid strategy's goals. It ensures that UK science takes a leading role in addressing the challenges faced by developing countries while also developing the UK's ability to deliver cutting-edge research and innovation (R&I) for sustainable development.

GCRF is overseen by the Department for Business, Energy and Industrial Strategy (BEIS) and implemented by 17 of the UK's R&I funders, which lead on commissioning R&I to address development challenges. GCRFfunded teams in the UK partner with organisations in the Global South to deliver interdisciplinary R&I on a wide range of urgent issues, from health and well-being to peace and justice, alongside agile responses to global crises such as Covid-19.

The purpose of GCRF's evaluation is to assess the extent to which GCRF has contributed to its objectives and impact. This has a dual learning and accountability purpose, as clearly set out in the evaluation objectives:

- 1. To assess whether the fund is achieving its aims (accountability)
- 2. To assess whether it is on course to achieve impact (accountability)
- To support BEIS in their development of a cross-fund and fund-specific key performance indicator (KPI) framework to provide a robust measure of the fund's impact and value for money (VfM) (*learning and accountability*)
- 4. To provide evidence of what works and make interim assessments of VfM to feed into GCRF learning loops to improve the fund while it is in operation (*learning and accountability*)
- 5. To inform the design of a VfM case for future funds (*learning*).

As the evaluation has both accountability and learning functions, it will provide evidence of GCRF's contribution towards impact and engage with BEIS's developing processes for learning about aid effectiveness.

Given the complexity of the fund, the evaluation is designed in three stages from 2020 to 2024. The evaluation design was developed under the earlier Foundation Stage Evaluation carried out in 2017–18.³ It addresses the purpose through five MEQs and a three-stage design that tracks GCRF's ToC from activities to impact over five years. Each stage applies specific modules to focus on different aspects of the ToC and the fund. Stage 1a of the evaluation runs from May 2020 to February 2021. The first stage – Stage 1a – consists of four modules conducted in parallel that aim to understand how BEIS and GCRF's delivery partners (DPs) manage and position the fund to deliver on its intended aims and commitments. These four modules focus on GCRF's management, relevance and targeting, fairness and the integration of gender, and social inclusion and poverty in the fund's commissioning and processes.

1.1 Strategic and policy context for the GCRF GESIP audit in 2020/21

The legal and policy context for GESIP is framed by three UK acts. First there is the International Development Act (2002), with an explicit requirement that the provision of ODA can show it is likely to contribute to poverty reduction.⁴ Second, since 2014, UK law requires ODA spending 'to contribute to reducing inequality between persons of different gender'.⁵ Initially the legal requirement applied to the now defunct Department for International Development (DFID), and by 2017 it extended to BEIS and GCRF. Beyond legal compliance to ODA spending commitment, the R&I sector, including BEIS and its DPs, are subject to the wider UK equalities framework driven by law and policy, which also impacts on research development and delivery. Third, the 2010 Equality Act requires that public organisations⁶ promote equal opportunities for everyone and protect the rights of those with protected

³ GCRF, 2018. GCRF Evaluation Foundation Stage. Available at: <u>https://www.gov.uk/government/publications/global-challenges-</u> research-fund-gcrf-foundation-stage-evaluation

⁴ UK Parliament International Development Act 'may provide any person or body with development assistance if he [*sic*] is satisfied that the provision of the assistance is likely to contribute to a reduction in poverty'. Available at: <u>www.legislation.gov.uk/ukpga/2002/1/contents</u>

⁵ UK Parliament International Development (Gender Equality) Act (2014). Available at:

www.legislation.gov.uk/ukpga/2014/9/section/1/enacted

⁶ UK Parliament Equalities Act (2010). Available at: <u>https://www.legislation.gov.uk/ukpga/2010/15/contents</u>. There are nine protected characteristics: age, disability, gender reassignment, race, religion or belief, sex, sexual orientation, marriage and civil partnership, and pregnancy and maternity. All of these were covered by protected by previously existing legislation which the 2010 Act replaced.

characteristics (including but not limited to gender, sexual orientation, age, race and disability) through implementing policies and strategies which prevent disadvantage and discrimination.

These legal accountabilities set the parameters of GCRF as a government ODA fund, which expects that funded research show development impacts that are poverty focused, gender sensitive and socially inclusive. As such it is incumbent on the R&I sector to show ODA compliance through tackling instances of under-representation, differential needs and systemic disadvantage to improve the relevance of R&I findings, to support inclusion, to reduce the impact of bias and, ultimately, to contribute to reducing poverty and thereby inequalities between genders.⁷

Moreover, the imperative to address Equality, Diversity and Inclusion (EDI) quickened in 2020, with Covid-19 and 'Black Lives Matter' throwing a spotlight on deep and persistent inequalities which are now receiving increased attention in the R&I sector. Many of GCRF's DPs have in consequence signalled increased priority to EDI. ODA gender and poverty requirements under the 2014 Act remain in force although, with substantive changes in ODA architecture and policy context under way, the strategic and policy framework for gender and other equality commitments is also subject to possible reconfiguration.

The changing strategic and policy context for GCRF⁸ outlined in the evaluation's Inception Report (2020) is set to accelerate in 2021. Since GCRF started in 2015, the national policy discourse on ODA has been evolving. Several significant changes have taken place since 2020, with the implications for GCRF still emerging.

First, in February 2020, the UK government announced the Integrated Review of foreign policy, defence, security and international development.⁹ This review covers all aspects of the UK's place in the world, from the role of the diplomatic service and approach to international development to the capabilities of the armed forces and security agencies. (At the time of writing in early 2021, the Integrated Review has not yet been published.) The emerging vision is to achieve influence in an increasingly complex world by bringing together all of the UK's national assets in a coherent, fused approach.¹⁰

Second, the merger in August 2020 of the Foreign & Commonwealth Office (FCO) and DFID into the new Foreign, Commonwealth & Development Office (FCDO) is expected to be central to the delivery of this emerging vision. It is anticipated that the broad view of national interest will be based on values (e.g. open societies and democratic values) as much as it is on the UK's core interests of security and prosperity. In November 2020 the Foreign Secretary announced a new strategic framework for ODA that will replace the UK government's 2015 aid strategy.¹¹ The framework notes the lack of 'coherence, oversight or appropriate accountability across Whitehall' for aid spending. The new framework sets out a range of measures to deliver better outcomes, including focusing aid on seven global challenges, focusing on countries where the UK's development, security and economic interests align, and increased oversight by FCDO of

⁷ Cavaghan, R. (2020) Gender, Science, Technology and Development: Literature Review for the University of Edinburgh's Working Group on Integrating Gender in GCRF Applications. University of Edinburgh, January 2020, unpublished.

⁸ GCRF Evaluation, Inception Report, Itad/Rand, August 2020, unpublished.

⁹Prime Minister's Office, 2020. 'PM outlines new **review** to define Britain's place in the world' [press release]. Available at: <u>https://www.gov.uk/government/news/pm-outlines-new-review-to-define-britains-place-in-the-world</u>

¹⁰ This may be influenced by the **fusion** doctrine. HMG, 2018. National Security Capability Review, March 2015. Available at: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/705347/6.4391_CO_Nationa</u> <u>I-Security-Review_web.pdf</u>

¹¹ 'UK aid to refocus on countries where 'interests align', Devex, 25 November 2020. Available at: <u>https://www.devex.com/news/uk-aid-to-refocus-on-countries-where-interests-align-98648</u>

ODA allocations to other departments. Programmes will be judged by fit with the UK's strategic objectives, evidence of impact achieved and VfM.¹²

Alongside strengthened FCDO oversight of ODA spend and the Integrated Review, the Covid-19 pandemic is also likely to influence broader policy changes taking place to ODA spending and management – and perhaps more so than any other time in the last 30 years. The economic recession and resultant fiscal policies have affected the Spending Review that was carried out in autumn 2020, limited to a one-year time frame and featuring a reduction in the ODA commitment from 0.7% to 0.5% of gross domestic product (GDP).¹³ New legislation is planned to reconcile this decision with the 2014 International Development Act, but it is not clear how this will relate to the 2002 International Development Act, which binds UK aid to make a 'contribution to a reduction in poverty'.¹⁴ Nor is it clear what the implications of this are for GCRF funding at the time of writing.

In the research sector, the formation of UK Research and Innovation (UKRI) in 2018 brought a shift in how the nine Research Councils (RCs) operate. UKRI was created to strengthen crossdisciplinary research and collaboration. UKRI's International Development Team provides central leadership and capability on GCRF strategic management and evaluation functions, managing many of GCRF's large investments centrally from the International Development Team, in collaboration with the individual Research Council. GCRF's overall fund management function, while part of BEIS, is also hosted within UKRI, creating a centre of gravity for international development research.

In 2020, Covid-19 has had an impact on research institutions and especially universities, both in terms of budgets¹⁵ and capability. This will not only affect the delivery of the evaluation but will also change the strategic context where the purpose of GCRF may be modified, including the ways commitments to GESIP are integrated into the ecosystem.

Taken together, this shifting context is likely to have significant impacts on GCRF's strategic role, funding and objectives, including those relating to GESIP, during the evaluation period. For the period under audit there are lessons for the future on how GCRF, with legal ODA accountabilities in relation to GESIP, addresses the inherent creative tension at its core between meeting the demands of ODA-supported research while still respecting the independence of researchers enshrined in the Haldane principle,¹⁶ the dynamics of which will be further tested by the rapidly changing context for ODA research in the UK.

1.2 Addressing GESIP within the GCRF: key principles and findings of previous GESIP assessments

At its inception, GCRF identified four key principles which guide how it supports UK and international researchers and innovators to take on key issues affecting developing countries. These principles state that all GCRF-supported research must demonstrate:

¹² Letter from the Foreign Secretary to the Chair, International Development Committee, 2 December 2020. Available at: <u>https://committees.parliament.uk/publications/3683/documents/38142/default/</u>

¹³ Insight. Spending Review: Reducing the 0.7% aid commitment. House of Commons Library. Published Thursday, 26 November 2020. Available at: <u>https://commonslibrary.parliament.uk/spending-review-reducing-the-aid-commitment/</u>

¹⁴ 'Poverty reduction missing from new UK aid strategy', Devex 26 November 2020. Available at: <u>https://www.devex.com/news/poverty-reduction-missing-from-new-uk-aid-strategy-98655</u>

¹⁵ In April 2020 the sector-wide loss **from** tuition fees to universities was estimated at £2.6 billion. London Economics, 2020. Impact of the Covid-19 pandemic on university finances, April 2020. Available at: <u>http://londoneconomics.co.uk/wp-content/uploads/2020/04/LE-Impact-of-Covid-19-on-university-finances-FINAL.pdf</u>

¹⁶ The Haldane principle, originally stated in 2018 and reinstated by a ministerial statement in 2010, holds that decisions on individual research proposals are best taken by researchers themselves through peer review.

- a challenge and impact focus
- interdisciplinary research excellence
- global and equitable partnerships
- organisation and leadership.

GCRF has provided extensive guidance¹⁷ on these principles and how they underpin the goal and objectives of the fund. Absent from these materials is explicit discussion and guidance on the fund's gender, social inclusion and poverty ODA commitments.

The absence points to an underlying challenge to the framing of ODA-supported research which has been picked up in three previous reviews: the Foundation Evaluation of GCRF, the BEIS Gender Equality Review of GCRF and Newton Fund, and external evidence reviews conducted by UKRI on the key challenges for EDI in the UK and internationally.

These reviews highlighted that **GCRF's policies and systems did not directly address GESIP issues at the inception of the fund, although all three dimensions are implicit in its objectives and goals.** All indicate an increasing focus on gender equality from 2018 onwards both from GCRF and the wider R&I sector. To a lesser extent, there was increased attention to social inclusion in terms of organisational diversity. The internal reviews point to a number of blockages in the way GESIP issues are addressed in the *planned* outcomes and pathways to impact of GCRF-supported research.

The challenge for GCRF, as the reviews indicated, is that without embedding GESIP within systems management as well as the design and implementation processes shaping funded research, these blockages undermine the effectiveness of GCRF. Without a strong and strategic GESIP focus, the development impacts and ODA excellence expected by GCRF through its principles, goal and objectives are unlikely to be met, as the findings from the reviews detail.

The GCRF Foundation Stage Evaluation¹⁸ in November 2018 highlighted weaknesses in GCRF's approach to gender equality and pro-poor socially inclusive research. As a result, the report provided a co-produced ToC, with one of the intended impacts being:

'contribute to achievement of the SDGs [sustainable development goals], enhancing people's wellbeing, improving equality for people of all genders, promoting social inclusion, economic development and environmental sustainability in developing countries'.

Underpinning the pathways to impact is the explicit assumption that gender and social inclusion can be designed into R&I for inclusive impacts, generating both short-term and longer-term outcomes on the way to impact.¹⁹ A necessity for GCRF in achieving the intended impact, emphasised by the Foundation Stage Evaluation, is bringing GESIP into the evaluation purpose of the fund and setting a baseline through a formative audit.

¹⁷ Through its website, face-to-face seminars, webinars and online videos. See for example: <u>www.youtube.com/watch?v=Ow0y05-Kh5g</u>

¹⁸ GCRF, 2018. GCRF Evaluation Foundation Stage. Available at: <u>www.gov.uk/government/publications/global-challenges-</u> research-fund-gcrf-foundation-stage-evaluation

¹⁹ Ibid. (Part A Theory of Change – Figure 3 Theory of Change, pp. B1–3).

The UKRI series of reviews in 2019 on EDI found that the focus of research internationally was still largely at the nascent stage in addressing intersections and interlinkages between different dimensions of inequality and discrimination.

'A large proportion of interventions focused on gender (or sex) equality, or wider EDI issues (such as 'diversity training'). Other characteristics such as disability, religious inclusion or age were less likely to be the primary target of interventions'.²⁰

Importantly, the international review of EDI approaches concluded that monitoring and evaluation exercises (particularly those focusing on effectiveness) were constrained by a lack of comparable data sets across different RCs. The review also pointed to a wider issue in commissioned international research, of which GCRF is part; that is, the emphasis has been on EDI, including gender diversity, in the management and composition of research and research teams. Less (albeit increasing) attention has been given to the content and impacts of research. In 2019 UKRI, as part of its EDI commitments under the 2010 Act, also published data for diversity characteristics of its funding applicants and recipients from 2014 to 2019, including (post-2016) GCRF data.²¹ The data indicated positive changes in composition of applications for women and ethnic minorities, with no change in composition in applicants with disability. The data also indicated that large awards and the role of principal investigator (PI) remained dominated by white male applicants. A second review is scheduled for 2021. There is potential in the future for disaggregating GCRF data from the main data set for comparative and stand-alone analyses.

The implications for GCRF are twofold. First, the findings highlight the synergies and reinforcing drivers for strengthening attention to equality, including gender equality, and inclusion between the GCRF and the wider UK R&I sector. Second, the existing systems do not enable the R&I sector to capture and amplify the results and impacts of ODA-support research, resulting in missed opportunities to capitalise on the synergies to improve the gender and inclusion dimensions of the fund's operations.

The BEIS Gender Equality Review of GCRF and the Newton Fund, completed in April 2020 just prior to the start of this evaluation, found that BEIS was committed to improving its approach to gender equality as an administrator of ODA funds. In response to critical reviews, efforts have been made to strengthen the approach. It was noted that these were largely focused at the DP level, with less attention given to strengthening the overarching architecture and systems and strategies of the GCRF with respect to gender equality, diversity and inclusion (GEDI).

The review did not address broader aspects of EDI, the extent of gender mainstreaming within BEIS and its DPs, or the intersection with ODA poverty commitments. Nor, since it was outside its remit, did the review assess in detail the processes, content and impact pathways at the award level. It did note, however, the mixed picture at the award level in how gender equality and inclusion were being addressed because of the absence of an overarching strategy guiding BEIS's approach.

²⁰ Moody, J. & Aldercotte, A. (2019) Equality, diversity and inclusion in research and innovation: international review, Advance HE, for UK Research and Innovation. London, UK.

²¹ UKRI (2019) Diversity results for UKRI funding data 2014-15 to 2018-19. It analysed data from UKRI and the Arts and Humanities Research Centre (AHRC), the Biotechnology and Biological Sciences Research Council (BBSRC), the Engineering and Physical Sciences Research Council (EPSRC), the Economic and Social Research Council (ESRC), the Medical Research Council (MRC), the Natural Environment Research Council (NERC) and the Science and Technology Facilities Council (STFC).

The recommendations (see Box 2 below) focused on the management of GCRF to improve effectiveness. They were predicated on the assumption that addressing gender, diversity and inclusion systematically, coherently and across all levels of GCRF would increase the impact of the fund, resonating with the ToC developed through the Foundation Stage Evaluation. BEIS intends to use the report to inform the development of its ODA R&I GEDI Strategy. Responses to the recommendations are a work in progress within BEIS, with the expectation that findings from this audit will provide a further steer in how to develop a strategy that strengthens its approach to GESIP.

Box 2. Review of GCRF and Newton Fund evaluation (2020)

The Review of Approaches to Gender Equality in GCRF and the Newton Fund made four interlinked recommendations aimed at strengthening the management, operations and monitoring, evaluation and learning (MEL) of both funds:

- Recommendation 1: BEIS should develop a clear, coherent approach to gender equality, diversity and inclusion at the ODA portfolio level.
- Recommendation 2: BEIS should institute a systematic approach to learning, clearly differentiating between internal and external functions at key stages in the fund management cycle.
- **Recommendation 3:** BEIS should develop a fund-level approach to gender equality, diversity and inclusion that is tailored to each of the funds.
- Recommendation 4: BEIS should ensure there is dedicated senior management capacity, sufficient resourcing and clear accountabilities at all levels to design, implement, monitor and adapt the approach.

None of the reviews directly assessed how the GCRF addressed poverty as an ODA requirement through the layers of operation and by the different entities under the fund – the 17 DPs, the Hubs and the challenge areas. The finding from the Foundation Stage Evaluation that the pro-poor focus of the GCRF was not well defined is picked up by the audit and addressed in detail in EQ 2.

These three reviews provide a good baseline from which to explore the broader aspects of GESIP across the GESIP ecosystem. The evidence indicates that there are extant processes and practices that, if harnessed, can support the GCRF to strengthen its approach to GESIP and move from ODA compliance to ODA excellence. Yet the findings also underscore the challenges the GCRF has faced and still faces in developing a strategic focus for GESIP in a complex ecosystem when the original architecture for the fund and its founding principles did not support GCRF to embed GESIP in its systems and policies.

1.3 Structure of the report

The GESIP audit report is structured as follows. After this section, section 2 outlines the audit methodology. Section 3 presents the findings against three of the sub-EQs (sub-EQs 1.5, 1,6, 1.9). Section 4 gives the conclusions, broadly structured around sub-EQ 4. Finally, section 5 outlines a set of recommendations designed to support the GCRF strengthen its approach to GESIP.

2 Methodology

This section provides an overview and justification of the methodology used in the GESIP audit. First, we give an overview of the GESIP audit (which includes the audit principles and framework) second, we give a summary of the data collection and the data analysis approaches. The section ends with a review of the strengths and limitations of the audit approach and associated methodology.

2.1 Overview of the GESIP audit approach

The approach taken in the GESIP audit takes into account the previous reviews of GESIP issues in the context of GCRF and ODA-funded research (see sections 1.1 and 1.2 above). The audit therefore builds on an understanding of how well GESIP are taken-up across the GCRF ecosystem, where progress has been made, where there are ongoing blockages, and where there are opportunities to further improve the GESIP compliance focus of GCRF.

The audit combines two audits set out in the original Terms of Reference (ToR): the formative poverty and social inclusion audit and the gender audit. These were consolidated, firstly because of the important linkages between the two in promoting ODA compliance and excellence, and secondly to optimise efficiencies in data collection and analysis, thereby reducing the burden on DPs and award holders.

In summary, the audit approach focuses on:

- where and how strategies, policies and processes have provided a consistent strategic focus for GESIP
- what expertise is available to support award holders integrate GESIP issues into their research
- how far MEL activities are supporting improvements in capturing the GESIP dimensions of research outcomes and pathways to impact and,
- the extent to which the GESIP dimensions of supported research are being communicated and disseminated across GCRF and externally.

2.1.1 Principles and framework of the audit

The audit takes a forward-looking, appreciative and utilisation-focused approach that recognises the embryonic state of current GESIP strategies, policies and processes across the GCRF ecosytem. It thereby covers both the accountability and learning interests of GCRF.

Mainstreaming gender into GCRF is ongoing. The audit, in recognition of significant action taken to strengthen its approach to gender, assesses how far GCRF stakeholders are meeting ODA gender accountabilities. It also focuses on the wider context of GESIP in R&I and how, if at all, stakeholders have considered the impacts of Covid-19 on the GESIP dimensions of research. Further details of the approach are given in Annex 1.

The audit framework is underpinned by selected components from the McKinsey 7s framework,²² together with a Social Relations Framework²³ (see Annex 1). While the McKinsey 7s framework provides the framing to review the GESIP categories from an 'institutional mainstreaming' perspective, the adapted Social Relations Framework adds emphasis on more context-specific categories of analysis that relate specifically to GESIP concerns. The framework is naturally reliant on specific understandings of GESIP and audit terminology (see Box 3).

The components of the frameworks are embedded in the audit's evaluation matrix (see Annex 2). Each EQ was addressed using a harmonised set of judgement criteria relating to strategies, processes, expertise, MEL and communication, plus the issue of accountability for exploring compliance for mainstreaming GESIP issues. The audit also included a particular interest in the presence and reporting of disability-sensitive approaches, as well as in the implications of Covid-19 adaptations.

Box 3. Audit, gender, social inclusion and poverty definitions

An audit is an independent assurance activity designed with an objective to add value and improve the working and operations of a particular organisation. An audit approaches differs from an evaluation approach in that the latter is often guided by OECD/DAC criteria,²⁴ while the former generally tend to prioritise compliance interests along a predetermined set of guidance materials at the expense of wider contextual enabling environment concerns. Consequently, audits tend to generate findings more aligned with accountability concerns – although the GESIP audit also incorporates a strong focus on appreciate learning.

Gender is a social scientific term used to describe shared social ideals of femininity and masculinity, associated behavioural expectations and relations between sexes.²⁵ These shared ideals vary across organisations, countries and legislative and cultural boundaries. Gendered attitudes and behaviours are (re)produced at multiple levels – in individuals' identities and expectations, in social, economic and political institutions, and in wider society and culture. This leads to inequalities between genders, where one gender (usually women) is excluded or disadvantaged in terms of accessing and benefiting equally from economic, political and societal resources and opportunities. The concept of 'Intersectionality' – an important component in understanding gender equality – draws attention to the ways in which different identity markers (e.g. disability, race age, migration status, caste) intersect with each other to structure privileges and disadvantages. It incorporates the idea that gender, social inclusion and poverty can combine to further exclude gender identities.

Social inclusion (SI) is defined as the process of improving the terms of participation in society, particularly for people who are disadvantaged, through enhancing opportunities, access to resources, voice and respect for rights.²⁶ Some authors focus on the three domains of social participation, economic participation and political participation as key aspects of SI, thereby capturing elements around gender equality and poverty.²⁷ However, for the purposes of this

²² www.mckinsey.com/business-functions/strategy-and-corporate-finance/our-insights/enduring-ideas-the-7-s-framework. See the Management Review module for a fuller explanation of the 7s framework.

²³ Kabeer, Naila, and Ramya Subrahmanian. Institutions, relations and outcomes: Framework and tools for gender-aware planning. Brighton: Institute of Development Studies, 1996.

²⁴ <u>www.oecd.org/dac/evaluation/daccriteriaforevaluatingdevelopmentassistance.htm</u> accessed February 2021.

²⁵ Cavaghan, R. (2020) Gender, Science, Technology and Development: Literature Review for the University of Edinburgh's Working Group on Integrating Gender in GCRF Applications. 10.13140/RG.2.2.18199.73125. Further elaborated in Tetra Tech (2020) The Newman Fund and The GCRF: Review of approaches to gender equality, p.7.

²⁶ UN Department of Economic and Social Affairs (2016) Report on the world social situation 2016. Leaving no one behind: the imperative of inclusive development. Available at: <u>https://www.un.org/esa/socdev/rwss/2016/chapter1.pdf</u>

²⁷ Beall, J., & Piron, L. H. (2005). DFID social exclusion review. London: Overseas Development Institute, 32, 37.

evaluation we will exclude gender and poverty from our definition of SI and will also highlight aspects concerning inclusivity issues for disabled persons.

Equality, diversity and inclusion (EDI) is a conflagration of terms closely linked with that of SI, and is common parlance in the GCRF ecosystem (see Guyan & Oloyede (2019) and Moody & Aldercotte (2019)). The use of the EDI acronym often implicates gender concerns with respect to 'equality' – and can be used as a catch-all for gender equality and SI concerns. However, for the purposes of this report, SI/EDI is combined, and excludes a focus on gender equality (GE).

Gender equality, diversity and inclusion (GEDI) refers specifically to the upcoming GEDI strategy being developed across GCRF. It should not be conflated with SI or EDI for the purposes of this report.

Poverty entails more than the lack of income and productive resources to ensure sustainable livelihoods. Its manifestations include hunger and malnutrition, limited access to education and other basic services, social discrimination and exclusion, as well as the lack of participation in decision making.²⁸ We take a consciously broad approach to the framing of poverty in this note in order to capture the full range of ways in which it is expressed in and across GCRF structures.

2.1.2 Scope of the audit

The successful integration of GESIP into a fund as complex as GCRF (see Figure 1) requires that, as the BEIS Review of Gender Equality recommended, a clear direction is set at the portfolio level and that dedicated capacity and clear accountabilities cascade down the different levels of the fund, with mechanisms in place to allow learning, monitoring information and results to percolate upwards. With these conditions in place, GESIP concerns can be designed into innovative research addressing development challenges for inclusive impacts.

²⁸ <u>https://www.un.org/en/global-issues/ending-poverty</u>

Figure 1: Overall structure and allocation of GCRF funds



19

The management module noted that the complexity of the fund – with over 30 transition points where funding moves from one resource holder to another – provides a challenge for 'finding a strategy, structure and system that can sustain a consistent focus on development impact'.²⁹

As a consequence, the GESIP audit focused on a simplified version of the GCRF structure to frame the audit scope of enquiry. For instance, the GCRF can be presented as having four levels (see Figure 2). For the GESIP audit, this scoping emphasises the dynamic linkages across the GCRF system – in particular:

- how well the current and developing policies, strategies and processes are cascading down from primarily the DPs (level B) and also, to a lesser extent, GCRF (level A), and how well this supports the integration of GESIP into programmes (level C) and awards (level D)
- how GESIP results, learning and practices are being used to support better practice and development outcomes both upwards to level A (BEIS) and downwards to level D (the awards).



Figure 2: GESIP within different levels of GCRF management

Our scoping also reflects issues identified in earlier reports – including the Tetra Tech (2020) Gender Equality report.³⁰ These emphasise the importance of an overarching gender, inclusion and diversity strategy (developed in parallel with an approach to MEL) which is sensitive to these institutional dynamics. However, it is unrealistic to expect, in a fund as complex and multi-faceted as the GCRF, that any overarching strategy will work perfectly. As such, the heterogeneity and variety of the fund creates opportunities to learn from organic experimentation and adaptations regarding how GESIP can be successfully integrated into development research. The audit scope therefore appreciates that creating a collective system-wide approach to learning capitalises on the creative tension between independence of research and the strategic oversight of the fund. In so doing, the audit seeks to

²⁹ Management module report 2021:18.

³⁰ Tetra Tech (2020) Approaches to Gender Equality in the Newton Fund and GCRF.

appreciatively examine spaces where GESIP practices and innovations by researchers can be valued and adopted to support progress across the ecosystem.

Using this scope, and as outlined in section 2.2, the audit examined compliance on GESIPrelated **strategies**, **policies** and **processes**, **expertise**, **MEL systems**, **and communication** and **dissemination** within and across DPs and award holders.

2.2 Overview of data collection

Data collection for this GESIP audit was collected in four steps: a DP-focused literature review, an overarching award level topic assessment, an award-level audit and co-creation workshops. Data collection included interviews and extensive document analysis (see Figure 3).

Step 1 DP review Review of GCRF and DP policies & systems relating to GESIP at 'level A' and 'level B' in the award systems. Document review and interviews	Step 2 Assessment of award topics Topic modelling of GESIP keywords (including disability) from a curated dataset of core award data from Gateway to Research, ResearchFish and the BEIS GCRF tracker to understand how well GESIP issues are incorporated into documents	Step 3 Award-level audit Aim to take snapshot of up to 65 awards from a GESIP-focused and common sample set. Review of award cycle from proposal to output stage to rate how awards are addressing GESIP issues (from 'absent' to 'exemplary)	Step 4 Co-creation workshops Ground-truthing of key findings via two learning workshops with award holders, as well as discussion of initial recommendations
---	--	--	--

Figure 3: Data collection components of the GESIP audit

Restrictions resulting from Covid-19 and initial findings from the organisational and awardlevel review resulted in adjustments in steps 1, 3 and 4, which are described in Table 1. The core reason for this shift is that it became apparent in the inception stage that GCRF (level A) and DPs (Level B) were not recording information about GESIP consistently or systematically. In order not to overlook the evolutions in addressing GESIP over time, changes were made to the organisational assessment and the sampling criteria for the award-level audit (see Table 1 for details), while Table 2 provides a summary of the audit sample size against the four steps of data collection.

Method	Original	Adjusted
DP Review	Light touch document review at level A and B – BEIS and DPs	Extended document review focusing on DPs in order to further capture the dynamics between levels B, C and D
Award-level audit	65 GESIP awards randomly sampled across	Discretionary sample of 38 awards purposively sampled to capture

	DPs, challenge areas, Hubs and calls against sub-EQs	diversity of GESIP approaches and to optimise learning
Learning workshops	Up to six half- day workshops involving up to 60 people	2 online half-day workshops with restricted number of participants (12) to compensate for Covid-19 working context

Table 1: Summary adjustments made to original GESIP a	audit
---	-------

Method	Sample
DP Review	Analysis of 313 documents from 16 DPs (one DP had no documents with GESIP content). Special focus given to DPs which had awards being audited in the GESIP snapshots. Types of documents reviewed included: funding calls, proposals, strategic plans, delivery plans, funding/proposal guidance, notes, research Hub documents, impact and monitoring reports, and examples of communications/ knowledge dissemination.
Assessment of award topics	The sample size for the BEIS tracker review was 2925 awards (2016– 20), while for the topic modelling the sample was restricted to Gateway to Research (n=1,143). These awards were reviewed with respect to key words relating to gender, SI, disability and poverty (see Annex 7 for keyword list). A further topic analysis of just over 500 awards using data from ResearchFish was placed into clusters based on the similarity between abstract and text title.
Award-level audit	Of the target of 65 awards, the GESIP module reviewed 54 in practice (an achievement of 84%). This included a sample of six 'GESIP-led' awards from the core sample, 17 from the 'non-GESIP- led' awards from the core sample, and a further 31 from the GESIP- focused sample. An approximate total of 107 persons were engaged in discussion (49% female, 51% male) – a female to male ratio of 0.9 (almost parity). A total of 201 award-level documents were reviewed and graded according to topic guides (see Annexes 3 and 4). The selection criteria included: collective cover for the time frame 2016–20; a representative mix of small (<£100k) and large (>£1mk) awards; coverage for a range of DPs, including Hubs and Challenge areas; and a geographical spread – Asia, Africa and Latin America.
Co-creation workshops	 2 workshops – 11 participants (6 women, 5 men) purposively selected as leaders in their field (PIs, panel members, Hub leader, co-investigators (CIs), challenge leader). 23 key findings were collated and organised across the evaluation matrix. A set of semi-structured questions was developed to facilitate discussions. Questions were also asked about the impacts of Covid-19 on participant's awards.

Table 2: Breakdown of audit sample against the four steps of the data collection

2.3 Approach to data analysis

The approaches to the data analysis naturally varied between the individual methods and sampling focus:

DP Review: Documents relating to the policies, strategies and management of awards across DPs were mined and coded for GESIP-related text. Data was then logged against the evaluation matrix (see Annex 6 for the methodology). Data was also extracted from the BEIS tracker with respect to the Gender Analysis Marker (GAM) and disability markers and reviewed against the evaluation matrix (Annex 2).

Assessment of award topics: The GESIP topic modelling drew out information about how GESIP, with a particular focus on disability issues, were being integrated into grants and publications. The data science approach mined publicly available data triangulated with data from the BEIS tracker. Annex 7 provides the list of keywords modelled.

Award-level audit: Checklists for interview and award review protocols were developed against the evaluation matrix to strengthen qualitative enquiry into the current position of GCRF DPs and award holders in relation to GESIP ambitions (annexes 3, 4 and 5). Data was assessed using a set of rubrics which rated both the level of GESIP sensitivity in the award and the confidence in the evidence to make that judgement.

Co-creation workshops. Workshops were held to test key findings and explore conclusions and recommendations arising from the findings. Participants were purposively selected as leaders in their field with a history of engagement with the GCRF.

2.4 Strengths and limitations of the approach

A number of strengths and limitations were encountered in the design and implementation of the audit, none of which significantly undermined the principles or purposes of the audit.

Strengths:

- The review period of the audit (2016–20) allows a clear picture of the changes in the way GCRF addressed GESIP since its inception and the momentum of progress made. Looking backwards across the whole GCRF ecosystem enabled us to draw insights on the different focal points where GESIP practices and innovations emerged and where policies and strategies were able to gain traction in promoting GESIP improvements.
- The analysis covers different levels of the fund and allows for deeper analysis of DP GESIP policies and strategies as well as GESIP-sensitive practices at the award level. By expanding the DP document review, the audit was able to develop a robust baseline of the heterogenity of DP policies and strategies against which we could explore the effectiveness of processes used to implement them. The purposive sampling with the GESIP-focused sample optimised the team capacity to identify where the DP processes held traction and where the awards were generating independently gender-sensitive, inclusive and poverty-focused mechanisms of utility to the wider GCRF ecosystem.

Limitations of the GESIP audit:

• The audit approach provides a partial picture. Given the scale and complexity of the fund, we purposively selected awards for learning and supporting the progress made to date within GCRF. However, this was a small subset of the ongoing research being supported given the need to engage with both documents and key informant interview (KII) respondents from each of the 54 reviewed awards (from a total sample of 2,925 awards between 2016 and 2020). Inevitably, these snapshots will not capture

the nuances and differences in the interplay between DPs, award holders and the broader structures of the fund, or capture the full picture of the changes over time. Similarly, in some cases where key informants at award level were limited to PIs – because CIs were not available – the audit may not have fully captured in the analysis how some awards adjusted their design and implementation approaches during the research process and the reasons behind adjustments, if any. Collectively these limitations may have skewed findings at a granular level, but given the extensive document review and award-level assessment, the GESIP team are confident of the findings generated.

- Variability in the quality and quantity of data available for audit. For the DP GESIP data, there was substantial variation in the quantity and quality of documents available for review. Inevitably for some DPs, and at the programme level (level C) with large research portfolios, there was more GESIP-related documentation. There were major gaps in how DPs record, assess and support award holders, partly because the audit was not able to analyse DP response documents which record their decision-making processes in response to award proposals. Fund-wide reporting systems also did not consistently provide data on the gender or inclusion dimensions of the research or highlight a poverty focus. Equally, we could not assess the full GESIP dimensions of research, since reporting systems for capturing the range of research product and results existing reporting systems do not always include reguirements to report against the GESIP dimensions (e.g. outcomes of fellowships and capacity building; documentaries, webinars).
- The audit can only partially access the impacts of recent GESIP policy changes because of the time lag between introducing a stronger GESIP focus and full implementation. The introduction of new policies in late 2018 means changes are still filtering through the systems of the GCRF and its partners. Inevitability it will take some time to see the impacts on working practices and the communication of results.

3 Results

This section provides the findings for EQs 1 to 4. Each section begins with a summary finding for each sub-question, and is followed with supporting narrative – first from the point of view of delivery partner experience, and second from the point of view of award holders.

3.1 EQ 1: How are gender mainstreaming activities currently being strengthened throughout the fund?³¹

3.1.1 Strategies, policies, systems and processes to address gender equality

Summary finding: At the DP level there has been an expansion of policies, strategies and processes to review GE compliance concerns – particularly as of 2018/2019 – but this growth appears to be sporadic and broadly uncoordinated. At the award level, there is ample evidence to suggest that overall experience in framing and addressing GE concerns in strategies, policies and systems is fairly technocratic (e.g. having a 'parity' focus), highly variable and largely dependent on the inclination of both the award subject matter and associated teaming arrangements and interests within awards. Nevertheless, the proposal stage is seen to be a critical fulcrum where award holders and DPs converge on dialogue to address GE concerns in research systems and processes. Moreover, at this stage there appears to be a broad distinction between gender-oriented and non-gender-oriented awards, in that the former generally considered guidance to be overly simplistic whereas the latter found the framing sufficiently broad and non-prescriptive.

At the DP level, there is a clear trajectory of growth on policies, strategies and processes to address and promote GE that has been growing across GCRF since 2018/2019. However, this growth is fragmented and uncoordinated, and the breadth, depth and focus of such mechanisms vary substantially across the DPs. This finding draws on the review of DP-level documentation, where GE discussions are more evident since 2019. In addition, according to a key informant at the DP level,³² UKRI introduced GE statements early in 2019 on their own initiative and were not directed and/or facilitated by GCRF in doing so. Rather, it was stated that the main drivers of the shift were a combination of the Independent Commission for Aid Impact (ICAI) Review (2015), the Gender Equality Act (2014) and internal appetite among European and United States (US) colleagues: 'we anticipated there would be [compliance measures] and we acted'.³³

In particular, and as noted in the Tetra Tech Review (2020), the flagship example of promoting compliance with the Gender Equality Act using ODA requirements is UKRI's

³¹ EQ 1.5 has been reframed for the findings section. The original question sought to focus on recommendations rather than findings. The original EQ is retained in the conclusions section.

³² KII with DP (Scoping stage).

³³ Ibid.

introduction in 2019 of a mandatory gender statement requirement for all GCRF calls. The requirement is supported by guidance.³⁴

This gradual growth is also observed in 2020, when UKRI gave guidance on the design of Gender Equality Statements. The UKRI's GE compliance process has also led to some institutions (Edinburgh University – see Box 4 – SOAS and Imperial College) developing guidance for applicants on how to satisfy the GE compliance requirement. Interestingly, although written for the purpose of assisting the application process, some of these guidance notes expand and deepen the understanding of GE formulated in UKRI's process, providing a further background and context on gender dimensions in the context of ODA. The guidance offers more detail on *why* GE is relevant to GCRF research (beyond ODA compliance) and *how* researchers can better integrate GE understanding/analysis into their work in ways that go beyond equal participation and impact (DP Review).

Box 4. University of Edinburgh – Gender Equality Toolkit

As part of the genderED project, the University of Edinburgh developed a Gender Equality Toolkit for GCRF work. The toolkit provides guidance on the GE statement required by UKRI. It lays out the GE statement assessment criteria, provides background and context on the ODA rationale, and highlights why GE is relevant to GCRF research. Case studies are also used to illustrate key points (e.g. the links between poverty and gender inequality).

To aid thinking about the degrees to which GE can be integrated into research, the toolkit borrows Oxfam's rubric for integrating GE, which ranges from gender blind to gender transformative. The toolkit therefore provides guidance on how to consider GE at multiple levels: (i) project conceptualisation; (ii) equal and meaningful participation in the project; (iii) gender in impact and dissemination; (iv) monitoring risks. Each stage is accompanied by practical examples for illustration.

Accompanying UKRI, the UK Space Agency (UKSA), MRC, the British Academy and the Royal Academy of Engineering (RAEng) are also shown to be developing policies and processes explicitly to address GE within the context of GCRF (DP literature review). For example, UKSA's internal governance for the GCRF-funded portfolio explicitly highlights the need to be ODA compliant, including through contributions to enhancing GE.

However, at the other end of the spectrum, several DPs demonstrated little evidence of developing substantial systems for engaging with GE concerns. Indeed, a significant number of DP strategy/policy-level documents (strategic plans, delivery plans) were effectively gender blind, with no specific engagement with or references to GE (DP Review).

Guidance from DPs on institutional gender-responsiveness generally tends to focus on internal systems (team parity, etc.) rather than on research content or associated methods. This suggests that while some ODA guidance is being pursued among DPs, the overall focus is one of 'compliance' rather than an interest in 'excellence' per se. For instance, outside of UKRI's GE compliance process, where DP policy and strategy on GE was well-documented, guidance is usually framed in terms of EDI. In this sense, GE is focused largely on equal opportunities and affirmative action. This was particularly the case with UKRI and the RCs, as well as with the British Academy. Other examples include EDI strategies and action plans (UKRI and MRC), equality and diversity specified as key strategic aims or goals (UKRI, British Academy and NERC), the development of an external EDI Advisory Committee (UKRI), a series of internal

³⁴ UKRI (2019) Gender Equality Statement Guidance. Available at: <u>https://www.ukri.org/files/research/gcrf/genderequality-</u> statement-guidance/

reviews of GE and ethnic diversity (UKRI and MRC), and the promotion of steps to ensure peer review systems and research panel committees are gender equal (UKRI).

Overall, while GE guidance is deepening and widening at the DP level, it is difficult to clearly determine the degree to which it has been developed in response to the UKRI GEstatement requirements. However, there are certainly indications that UKRI and a small selection of other institutions have been engaged in a cascading and/or collective learning process that has stimulated a shift in the interests of GE concerns across the GCRF. By providing this leadership and a clear policy framework, UKRI has created space through which institutional best practice leaders (genderED, etc.) have been able to horizontally promote engagement and understanding of GE concerns. Nevertheless, the broader picture is one of DPs administering their GCRF funding separately, employing varied individual measures to address GE.

At the award level, the picture is similarly patchy compared to the DP level in terms of the consistency of GE strategies, systems and processes being integrated into various operational activities – although a small number of award holders have moved from 'a culture of compliance' to one of 'excellence'. For example, of the 1,143 GCRF Grants in Gateway to Research analysed through the topic modelling analysis, only 1% suggested a high focus on gender (>40 keywords), 6% demonstrated a consistent focus on gender (>20 keywords) and 24% showed low focus on gender (5 keywords).³⁵ While the majority of gender results mapped against the sustainable health and well-being challenge area (given this area contained the largest number of awards), there are nonetheless promising signs that gender terminology is fairly present across the topic modelling sample.

There are also clear signs that a small proportion of awards are recognising GE concerns in both internal systems and research content – particularly among award Hubs and awards that are centrally situated in the social sciences (see Box 5).

Box 5. Award holders taking advanced steps for gender transformative approaches

For the handful of awards that held GE as a central focus, there is a strong tendency to refer to critical 'feminist approaches' and the need to incorporate an intersectional lens into both the awards' methodology and teaming arrangements by going beyond terminology focused on the category of 'women's interests'.³⁶ Such awards are seen to be undertaking a 'gender transformative' approach to understand GE concerns as they apply to research processes. However, feedback from these award holders also strongly suggests that they think of themselves 'far in advance' of current 'compliance' framing, and that they lean more towards aspirations of 'excellence'.

However, even for awards that are comparative leaders in undertaking gender mainstreaming in internal or thematic content, the drivers are largely shown to be internal and dependent on predisposed interests and experience rather than direct or indirect support from DPs. Almost all awards that exhibited a strong 'feminist' foundation in research design and process articulated in interviews that they based this on either the gendered nature of the award or the inherent qualities of the team in question (or both). This trend was also observed among awards that demonstrated a less intensive or radical approach to GE.³⁷

By contrast, internally driven gender mainstreaming activities were less clear among awards that were not overtly 'gendered' in terms of their thematic focus. Such award holders

³⁵ See Annex 7 for a list of GESIP-related keywords used in the topic modelling.

³⁶ G18, G4, G5.

³⁷ G19, G29, G31.

reflected that they were largely appreciative of reminders and guidance on gender mainstreaming requested by DPs – although this guidance was almost uniformly encountered at the proposal stage only.³⁸ Indeed, common to the relatively few awards (social science based or not) that explicitly encountered some form of strategic guidance on gender mainstreaming was the fact this this support was almost always concentrated at the proposal stage.

Consequently, the proposal stage appears to be a critical entry point for all award holders to be engaged by both DP panel reviewers as well as written guidance on gendered subject matter. Despite this fairly systematic engagement at the proposal stage, the guidance is nevertheless largely considered to be perfunctory and light touch in nature by the majority of award holders. On the other hand, in a small minority of cases, award holders expressed frustration and concern that guidance to support gender-responsive thinking at the proposal stage is overly simplistic and largely 'meaningless in terms of [understanding] impact, and meaningless in ways for our own working'.³⁹ The same award team suggested that a space to present and discuss more complex and contested ways of understanding relationships of power was not facilitated by UKRI-ESRC in the proposal stage, which has the ultimate effect of building and perpetuating gender stereotypes⁴⁰ – a sentiment that was echoed by two other award teams that were highly experienced in gender-responsive and feminist thinking.⁴¹

Finally, on the few occasions where systematic and comprehensive support on strategic gender mainstreaming was afforded to award holders by GCRF, the support is largely more evident among either award Hubs or a small minority of very experienced award holders that declared a significant degree of familiarity with the GCRF ecosystem. These awards were able to 'tap' tacit networks and engage in horizontal learning activities to promote gender mainstreaming in a more strategic fashion.⁴² This suggests that there are pockets of promising practice among a cadre of award holders that can be specifically targeted for review and learning purposes.

3.1.2 Expertise available to support gender equality

Summary finding: A number of mechanisms to develop and maintain GE expertise have been observed at the DP level, although these mechanisms are generally uncoordinated and variable in nature. At the award level, particularly those with a humanities focus, many respondents self-reported a satisfactory performance with respect to ensuring appropriate GE expertise was developed and maintained. However, a strong tendency for focusing on team parity was observed at the award level, at the expense of more progressive thinking about gender-related barriers and power issues. This thinking has broadly developed without a clear line of sight of DP expectations regarding gender expertise.

There is good evidence showing that mechanisms have been put in place across the majority of DPs to build and maintain GE expertise at the DP level and award level – although these mechanisms are variable and mostly uncoordinated. As evidenced in the DP Review, the teaming arrangements, skills and competencies around gender that are promoted across GCRF include: the creation of advisory teams that supervise how GE is considered at all levels; the

³⁸ CS24, CS6-8, CS16, G33, CS99.

³⁹ KII 1, G3.

⁴⁰ Ibid.

⁴¹ G2, G5.

⁴² G4, G5.

provision of training programmes for employees; the designation of 'equality champions' to ensure that GE concerns are embedded within DP systems; and the promotion of toolkits that outline ways for capturing and building appropriate gender expertise. More specifically, UKRI engages with an EDI External Advisory Committee that reports to the UKRI Executive Committee on EDI performance. Similarly, Innovate UK has created a team of experts that addresses EDI topics. UKRI has also established an International Development (Gender Equality) compliance process that provides guidance and training to UKRI staff, applicants, peer reviewers and panellists (DP Review).

Similarly, MRC provides equality and 'unconscious bias' trainings to managers and employees and has developed a network of 'equality champions' composed of employees that work with human resource teams as well as local trade unions to closely monitor gendered teaming arrangements internally and ensure support. A good example of how these practices have trickled down within the GCRF ecosystem is the uptake of the Gender Equality Toolkit developed by the University of Edinburghwhich has been made available to the other institutions in the consortium through the Scottish Funding Council (SFC). In addition to this, no evidence of such expertise-building mechanisms has been found among the other DPs. Finally, with some minor exceptions (such as the MRC trainings), most of the practices and resources that have supported the building of gender-responsive expertise seem to have been initiated in 2019 (DP Review).

At the award level, there is routine respondent feedback and evidence suggesting that considerations on appropriate gender expertise are driven internally and without a clear line of sight regarding expectations that might be associated with ODA compliance concerns. Almost all award holders note that concerns for team parity arrangements are intrinsic to their ways of working, although this is often framed as a parity issue at higher levels of the award (i.e. among PIs and CIs), and/or in terms of ensuring a teaming balance at the data collection level. In addition, any shortfalls or missed opportunities in improving the parity of teaming arrangements are reportedly raised and addressed at the proposal or interview panel stage.

While the majority of award holders consider parity concerns in isolation of any formal guidance, only a minority of awards undertake a sophisticated approach to understanding and addressing gender expertise in their activities. The latter award holders are largely able to go a step further in considering not only parity but also the gendered expertise of core team members and the systematic barriers that they might encounter across the academic arena – particularly for early career researchers in low and middle-income countries (LMICs). Fewer yet provide strategic plans or informal visions relating to team capacity building in the medium to long term – one exception being an engineering-focused award that not only proactively sought to fill gaps in gender expertise at the pre-proposal stage but also considered capacity building activities for partner organisations at national level as well as for women smallholder farmer organisations.⁴³

A key characteristic that differentiates awards that go beyond a parity focus and those which seek to take a sophisticated or strategic approach to ensuring gender-specific expertise are addressed in their awards is those that have an explicit top-level interest in gender issues as a result of their award objectives and/or individual team member interests. For instance, a number of award holders referenced predecessor projects and associated networks, where 'both PIs were on advisory committees for other GCRF projects', and who had access to 'two external evaluations who are gender experts'.⁴⁴ Another award, situated in a hub, had access

⁴³ CS55.

⁴⁴ G18.

to a hub-wide 'Gender Champion' from the outset, as well as a gender expert located at the Hubs' executive advisory board,⁴⁵ and continues to embed gender champions at several research site locations where required.

Overall, given the general focus on parity, and that award holders generally state that they 'are doing it all already', few respondents requested specific suggestions on advanced approaches to gender-response capacity building at the team level. However, one respondent noted a cautionary concern that even the provision of 'more advanced' or intensive guidance may not necessarily equate to effectiveness: 'if guidance was made available on this, it is likely that there would be very little power analysis undertaken, and this would not be grounded in 'everyday' implementation and practice – thereby giving the impression of change, rather than building on insights gained from the empirical aspects of [research delivery]'.⁴⁶ This insight serves as a reminder that progress on formal guidance for improving GE expertise across GCRF also needs to be accompanied by organisational culture shifts, to ground progress into the everyday thinking and practice of DPs and award holders.

3.1.3 Monitoring and learning in GCRF's approach to gender equality

Summary finding: At the DP level there is no consistent approach regarding genderresponsive MEL approaches, while a small number of promising practices are emergent – the BEIS gender tracker being a promising example (although the tracker is seen to be error-prone and overly simplistic). At the award level, the picture is similar in that awards broadly maintain their own ad hoc systems for capturing disaggregated data. With respect to learning, this is largely conducted in a horizontal, organic and opportunistic fashion across both DPs and awards, and is clearly the dominant mechanism for learning diffusion. There is also considerable inequality between DPs in terms of the amount of MEL data captured and actively used for reflection vis-à-vis award holders.

Guidance from DPs on Gender Equality Statements and/or other gender-responsive MEL mechanisms in proposal documents and associated guidance is generally lacking, although some notable exceptions exist. Examples of systematic good practices for addressing GE concerns include UKSA's M&E function, used to measure the impact of the International Partnership Programme (IPP) on GE at programme and project level (e.g. Monitoring and Evaluation (M&E) Handbook 2020), or the Gender Monitoring and Evaluation Guidance provided by UKRI to assess the performance of Hubs in terms of GE aspirations. Additionally, UKRI engages with the International Development Team that coordinates and delivers an M&E of Gender Equality Act Compliance processes. More widely, SFC has requested universities to provide information regarding GE compliance at three different levels: within research, within GCRF activities, and in the context of projects. Higher education institutions subsequently reported on their plans to enforce GE considerations overall (e.g. achieving and Athena SWAN Award⁴⁷) and within specific projects (DP Review).

While UKRI and UKSA have developed tools and guidance documents that provide a framework supporting MEL within GCRF, few mechanisms have been found among other partners. Of these few, evidence of gender-specific or gender-responsive MEL systems includes review templates that explicitly address GE at the programme level or request gender

⁴⁵ G30.

⁴⁶ G3.

⁴⁷ The Athena Swan Charter is a globally recognised framework supporting gender equality in higher education and research.

assessments of applicants (Innovate UK), reports on applicants' gender diversity (Royal Society (RS)) or less formal means requesting higher education institutions to provide GE statements (SFC). For the remainder of DPs, no evidence of formal MEL mechanisms for capturing GE at the award level have been found. This may largely be a result of institutional inertia, given that Gender Equality Statements largely gained traction in 2019 and 2020.

As noted from a DP-level key informant, the other main mechanism for tracking gender concerns is ResearchFish, although there are also plans for UKRI to introduce an option for institutions to include a 500-word case study focused on gender in their proposals. As noted in parliamentary (International Development Committee) feedback from the Newton Fund Review (2020),⁴⁸ there are plans on track to begin the onboarding process for the Research ODA (RODA) platform as from October 2020, thereby allowing users to begin reporting on gendered components of ODA compliance.⁴⁹ However, as noted in the parliamentary response to the Newton Fund report,⁵⁰ it should be maintained that there is 'no one size fits all', given that DFID (FCDO) uses a different model and prioritises other ODA objectives.

At the award level, one of the central MEL mechanisms for capturing ODA compliance is the GAM, captured in the overarching BEIS tracker. Guidance is provided in the BEIS tracker to explain the logic through which a score is provided by reviewers (see Box 6), and the results of the tracker – as of August 2020 – are provided in Figure 4 below.

Box 6. BEIS tracker guidance for capturing progress on gender equality commitments

An activity should be classified as GE-focused if GE is explicitly promoted in activity documentation through specific measures which:

(a) reduce social, economic or political power inequalities between women and men, girls and boys, ensure that women benefit equally with men from the activity, or compensate for past discrimination; or

(b) develop or strengthen GE or anti-discrimination policies, legislation or institutions. This approach requires analysing gender inequalities either separately or as an integral part of agencies' standard procedures.

(2 - principal, 1 - significant, 0 - assessed but not targeted, blank - not assessed).

In summary, the evidence shows that approximately 15% of awards have not been assessed to date, and that a significant majority have been given a '0' rating (assessed, not targeted). The proportion of scoring for 'significant' and 'principal' engagement on GE is 6.6% (150 awards) and 4.3% (65 awards) respectively. However, a review of all awards in the BEIS tracker (n= 2,925 awards) that reflect a strong emphasis on gender and/or women in their titles shows that at least 30 awards have assessed by the GAM to have a '0' rating.⁵¹ Consequently, there is good evidence to suggest that a notable proportion (up to 26%) 52 of gender-focused awards

⁴⁸ <u>https://publications.parliament.uk/pa/cm5801/cmselect/cmintdev/742/74202.htm</u> accessed February 2021.

⁴⁹ RODA is an internal submission service that reflects requirements appropriate for the R&I context. It is a hybrid mixture of client data submission for DPs to report actual/forecast spend and finance/programme management. A range of tools and platforms is available for viewing BEIS published ODA data, including d-portal, DfID's Dev-tracker and a newly-launched International Aid Transparency Initiative (IATI) data query builder.

⁵⁰ Ibid.

⁵¹ The latter awards include research to address 'Visualizing Gender-Based Violence in Graphic Awareness Campaigns in Nepal', 'Debating, Performing & Curating Symbolic Reparations and Transformative Gender Justice', 'Articulating Women: Interrogating Intersectionality and Empowering Women Through Critical Engagements' and 'Improving healthcare at the intersection of gender and protracted displacement amongst Somali and Congolese refugees and IDPs'.

⁵² There are 150 1-rating awards and 65 2-rating awards, making 215 awards. Given that there are at least 30 awards that appear to have been overlooked, this constitutes a potential MEL shortfall of 26%.

are being overlooked in current MEL systems – suggesting there is room for improvement on how the GAM is maintained and reviewed.



Figure 4: Count of GCRF awards obtaining gender rating (0–2), 2016–20

The GAM guidance can also be argued to be overly simplistic in that, there is no distinction afforded to the gender-responsive nature of awards on either a thematic content or teaming level. As a result, the BEIS tracker marker cannot currently distinguish between performance on the gender-responsiveness of internal teaming arrangements and associated capacity building activities, as opposed to the degree of gender-responsive framing inside award content.

There are also shortfalls in the gender-responsiveness of MEL systems at the award level: feedback and documentation from award holders shows that while the promotion, uptake and completion of GE statements has been improving since 2019, a considerable majority of award holders are unfamiliar with the need to use or be informed by these statements. Award-level respondents stated that they were largely agnostic about the shortfalls in genderresponsive MEL systems outlined above. The majority of those interviewed explained that they maintained their own ad hoc systems for capturing sex-disaggregated data (either in teaming arrangements or in data review). However, a small number of respondents – primarily those engaged in gender-responsive awards – were critical of the tools and mechanisms to monitor progress on GE mainstreaming. Broadly put, the latter award holders consider GCRF MEL systems to be overly simplistic and inappropriate for unpacking complex behavioural change, given that little space is afforded to a deep analysis of unequal power relations. However, these award holders did not venture further in suggesting concrete solutions in this regard.

3.1.4 Accountability for gender equality ODA requirements

Summary finding: There is a variety of accountability measures that have been put in place by DPs to frame compliance on GE mainstreaming, but evidence of implementation with award holders – especially following the proposal stage – is lacking. Thereafter, where GE statements (or similar) are sent up the delivery chain, feedback on their quality or content is largely absent. There are, however, pockets of promising practice among awards which go beyond a 'compliance' framing towards one of 'technical excellence' – but these examples are scattered and few.

Accountability processes among DPs and awards for ensuring GE mainstreaming are generally fragmented and uncoordinated by GCRF, but are concentrated at the proposal stage – particularly in panel review dialogues. One of the limited examples found in the DPlevel review is the GE Act Compliance process, developed by UKRI, that formally requires awards to report an end-of-year survey, which includes an assessment of the award's impact on GE. Also, some funding guidance documents, published in or after 2019, explicitly require applicants to incorporate GE in specific sections of the proposals (e.g. in 'Pathways to Impact' and 'Case for Support' documents). Similarly, UKSA and Innovate UK's proposal templates incorporate GE requirements. UKSA's templates include a separate section on 'impact on gender' and Innovate UK refers to GE as a requirement for participation under the International Development (GE) Act 2014. Research England (RE) has included the act in grant conditions for the GCRF funding stream it administers.

As noted above (3.1.3), given the sporadic and embryonic MEL systems currently in place for reviewing ODA compliance, the formal upward accountability process for this is equally patchy and inconsistent. Rather, the majority of award holders note that the proposal stage (and associated interactions with the review panel) is the primary locus for formal feedback and corresponding adjustments to be made on proposals.⁵³ However, a small number of award holders noted that while this accountability process was generally constructive, it was not always effective: 'The reviewers forced us to unpack [gender issues] in more detail. They also encouraged thinking on [alternative gender-specific enquiries]'. In so doing, they demonstrated an understanding 'far better than ours' around these complications [and]were 'spot on – which is not always the case'.⁵⁴

The framing of 'accountability' beyond the proposal stage is largely understood and practiced informally in terms of horizontal learning and other internal systems at both the award level and the DP level. This finding is also evidenced in feedback from a number of award holders who state that if Gender Equality Statements (or similar) are developed at any point in the post-proposal stage and sent up the delivery chain, there is often no feedback or reflection back towards award holders in order to nuance their approach on ODA compliance for GE. As a result, apart from a few key awards that demonstrated that strong internal incentives for ensuring accountability mechanisms are in place,⁵⁵ the accountability process is fairly technocratic and focused on compliance at the proposal stage, with little focus regarding ongoing compliance or technical excellence later in the research delivery process.

3.1.5 Gender sensitivity of communication strategies and research outputs

Summary finding: At the DP level, there is no central or core guide through which the gender-responsiveness of communications or other outputs can be reviewed. At the award level, there is a generally a strong commitment to bottom-up and co-created communication and dissemination approaches – although these commitments vary in nature, and the degree to which gender-responsive messaging is consistently mainstreamed in practice is relatively unclear.

⁵³ G17, G18, CS44, G29, G37, G38.

⁵⁴ G2.

⁵⁵ G2, G3, G4, G5.

At the DP level, there is a considerable heterogeneity in the amount and type of communication material found – particularly the extent to which gender-responsiveness is an issue of concern. The typical dissemination material found includes case study reports (SFC, Innovate UK, Research England), brochures (UKSA), slides (RAEng) and event feedback reports (UKRI). Among these, only Research England and Innovate UK case study reports regularly include GE considerations as part of programme activities.

Similarly, award holders could rarely unpack the mechanics of a gender-responsive communication approach, nor could they articulate the role of DPs in encouraging such approaches. The majority of award holders were able to demonstrate basic or limited accounts of their communications and dissemination approaches in Pathway to Impact statements and/or Case for Support (proposal) materials. These accounts are largely focused on the targeting of specific stakeholder groups and on explanations of any innovative multimedia approaches. One of the few key documents to address gender-responsive communication and dissemination at the award level was the 'Developing inclusive conferences – Best Practice Guide' (Chautard and Hann, 2019).

Despite the broader lack of guidance materials, a small number of awards articulated in fair detail the promotion of bottom-up communication approaches and a commitment to including a gender lens in all their communication materials.⁵⁶ However, these awards were largely predisposed to a gender-responsive communications approach, on the basis that GE is front and centre in their award profile. These award holders developed the approaches internally, and at no stage were they made aware of any guidance from DPs or GCRF on expectations for gender-responsive dissemination or communication approaches.

3.2 EQ 2: How effectively is poverty framed and understood within the GCRF ecosystem?

3.2.1 How is 'poverty alleviation' understood across GCRF?

Summary finding: Addressing 'poverty alleviation' is considered by DPs and award holders as implicit in the research funded by GCRF. However, it is rarely – if ever – explicitly defined in relation to ODA compliance. There is no high-level BEIS/GCRF guidance or consistency in guidance across the DPs with respect to poverty terminology, although – in a limited number of cases – programmes and calls have considered poverty in their framing. There is also a high degree of diversity around how award holders understand and apply poverty concepts in the research process.

At the DP level, guidance on poverty or the poverty requirements of ODA funding is either non-existent or limited to specific programmes or calls. Many award holders report being given limited guidance in relation to poverty. For instance, the pathways to poverty alleviation are not requested in any observed guidance. There is some limited evidence of poverty alleviation being considered in the rationale of programmes and in the funding calls (UKRI and ESRC), delivery plans (ESRC, RAEng) or as part of communication material (RAEng, UKSA). As pointed out by some KIIs, one of the challenges historically has been that the GCRF was not

⁵⁶ G2, G4, GR6, GR7, GR2.

under the authority of the International Development Acts (2002, 2014), and therefore explicit framing of policies and programmes was not a requirement.⁵⁷

As a result, it appears that the interpretation of poverty implications has been left largely to the proposal writers and award holders. Since late 2018, there has been growing awareness among some DPs that more could be done to guide, shape and capture the poverty dimensions of supported research, and there are expectations among some award holders that there will be more clarity regarding poverty terminology in the future. However, some DPs observed that there were limited resources to address the poverty dimensions to the depth they would like, and cautioned a gradualist approach to change.⁵⁸

At the award level, poverty reduction appears to be largely understood in terms of working in and collaborating with developing countries listed as ODA countries - and these understandings are largely framed at the proposal stage.⁵⁹ Evidence from interviews and award-level documents shows that consideration for poverty dimensions occurs most often at the proposal stage. For instance, proposals and Pathway to Impact statements are largely the only places where poverty considerations are addressed, and the framing is primarily associated with relevant SDGs and the GCRF challenge areas – rather than an exposé of the detailed poverty dimensions that would be addressed by the award.⁶⁰ While awards may not be systematically defining or making explicit references to poverty, the majority were clear that poverty was being indirectly addressed given the inherent nature of the research and the context (e.g. economic and social development, food insecurity, human rights or health inequalities). Despite this, few awards provided details in the documents regarding how these concepts would be addressed, the poverty status of targeted populations or how the impacts of the research would contribute to poverty reduction. The only resource found across the ecosystem to support awards to consider the poverty dimensions of their research is the Edinburgh University's GE toolkit, which includes case studies that illustrate the links between poverty and gender inequality.

In a small minority of awards,⁶¹ poverty was understood in a broader framework encompassing structural inequalities and multiple deprivations. These awards, often also demonstrating robust approaches to SI and GE, considered the framing of poverty alleviation in the GCRF to be overly simplistic or problematic, taking account of neither the changing ODA context of poverty alleviation nor the advances in conceptualisation of poverty. For these awards, poverty is embedded within their methodological framework and intersects with concepts relating to, for example, spatial disparities, disabilities, identities, marginality, power, gender and exclusion.

The range of ways in which poverty and poverty alleviation have been understood across DPs and award holders highlights the challenges involved in tracing the framing of poverty throughout the fund. Findings from the topic modelling (n=1143 from Gateway to Research) suggested that there was a potential bias towards econometric or income-based categories of poverty, and there is some evidence in documents to support this finding.⁶² However, the lack of explicit guidance for awards about how to frame or report on the poverty dimensions of the research meant it was impossible to determine the extent of the bias. In a large proportion of the awards audited, documentation available indicated causal links were implicitly assumed,

⁵⁷ DP12, Feedback from Learning workshop, DP3.

⁵⁸ DP9, DP12.

⁵⁹ Almost 80% of common pool and over 50% of core sample.

 $^{^{60}}$ Over 70% of common pool and more than 50% of GESIP-focused.

⁶¹ G2, G5, G8, G19, G18, G26, G27.

⁶² 1143 GCRF Grants in Gateway to Research analysed through the topic modelling analysis.
without a formal process or mechanism within the GCRF for these links to be consistently interrogated or tested across the organisational layers of the fund. As one award holder observed: 'Poverty is a packed concept that intersects with a number of other development concepts.'

The key finding regarding the framing of poverty across GCRF, therefore, is that the current approach does not allow for the consistent unpacking of, and rationale for, varied understandings of poverty. This represents a significant omission given the critical framing of ODA-funded research from a poverty perspective.

3.2.2 Expertise available to address poverty alleviation within the GCRF

Summary finding: Capacities, resources and skills to address and integrate poverty into the proposals and implementation processes of awards remain largely with individual award holders, given there are no expectations from GCRF to outline relevant expertise. There are signs of some cross-fertilisation of expertise between awards, but these efforts are not amplified by current mechanisms and systems within the GCRF.

The findings indicate that there is neither a requirement or expectation within the GCRF ecosystem to define or outline the expertise and resources available to address poverty issues. A small number of DPs also noted that ODA compliance was a key driver for assessing, reviewing and monitoring how poverty had been dealt within awards, including drawing on expertise to support the research approach.⁶³ Some KIIs reported valuing the poverty expertise in the peer review and selection process, although there was no clear evidence that poverty expertise was routinely brought into the call, appraisal and selection process.⁶⁴

Nevertheless, the vast majority of award holders reported undertaking a number of activities that indirectly informed ODA compliance concerns in relation to fulfilling 'poverty alleviation' expertise. How this was interpreted varied widely, however, from consulting with poverty experts in the design of research to having PIs and CIs with a poverty research background and to having a poverty expert in the term: 'We did not benefit from any assistance or guidance on poverty alleviation [but it] is implicit as a goal in the award since the focus in on communities in poorer countries.'⁶⁵ Despite these team activities and the availability expertise, poverty criteria were almost never included at the express interest of DPs, and award holders routinely declared that they were unaware of expectations in this regard.

Consequently, with the limited articulation of expectations on ODA compliance at both the DP and award levels, and with the variable interpretations of poverty in practice, there is limited scope for mapping and assessing GCRF's contribution to poverty alleviation. Given this constraint, the effective mapping of GCRF contributions to the SDGs is likely to be problematic.

3.2.3 Monitoring and dissemination of poverty data, findings and research outputs

Summary finding: Reporting requirements with regard to poverty-related approaches, outputs or outcomes are not consistent across the GCRF, resulting in a broken evidence trail. Data relating to poverty and poverty alleviation remains at the level of the DP, the

⁶³ DP3, DP5.

⁶⁴ Co-creation workshop, G2, G22.

⁶⁵ Award KII.

Hub, or in many cases the award holder. The are no systems and mechanisms in place to feed poverty data routinely upwards through the GCRF's organisational layers or to disseminate more widely.

At the DP level it has been challenging to identify the extent to which GCRF-supported research has contributed, or is in the process of contributing, to pro-poor outcomes. The audit found limited evidence that the poverty dimensions of awards were being systematically documented and disseminated. Although the GCRF ToC 'envisaged that a short-term outcome for the fund is the direct application of pro-poor practices, technologies and products as a result of participating in products', monitoring systems are not designed to capture the distinct poverty dimensions of research.⁶⁶ The main means for monitoring are programmes and project reports (the British Academy (BA), the Higher Education Council for Funding Wales (HECFW), the Academy of Medical Sciences (AMS), the Department for the Economy Northern Ireland (DEFNI) and Innovate UK). However, both DPs and award holders reported that current procedures did not allow for the full range of poverty-related results to be captured. There are limited avenues to report the multi-dimensions of poverty and the interlinkages with gender and SI.

The introduction of narrative reporting allowing for increased visibility of poverty issues (and all GESIP) in mid-2020 is seen as a positive step. There was poverty data available in some DPs, but mechanisms for elevating data up or throughout system were not in place. DPs reported lack of resources to do so and disincentives in the current system, where it was felt there were challenges in reporting the multi-dimensions of poverty and the interlinkages with gender and SI: 'There is an issue around how we are asked to report and what we report against [...] What you ask us to report against can determine what you think you are seeing [...] We frame [our reports] around what we are being asked. The granularity is lost and what can be learnt and developed from different perspectives'.⁶⁷

Beyond ODA compliance and 'Pathways to Impact' statements, no additional tangible guidance appears to have been provided to award holders with respect to the reporting requirement and MEL expectations in relation to poverty.

The findings are similar at the award level, where the absence of any requirements to monitor and report routinely on poverty suggests that poverty findings and results are being under-reported. Findings from the topic modelling data set indicated that 26% of the awards analysed in the Gateway to Research (GtR) data set had a consistent focus on poverty (20 or more terms found in the documents). An indication of a higher focus on poverty in reporting research results as compared to GE and SI. Yet the results showed a bias towards a few research clusters (notably maternal pregnancy, peace and violence , and civil society).

However, subsequent analysis of ResearchFish and the BEIS tracker, triangulated with the literature review and award audit, suggests three reasons why the picture is more complex. First, published research is not using specific poverty terminology to describe research activities, nor are poverty-related tags or categories being used routinely during reporting processes at the award level. Second, many of the awards disseminated their findings through workshops, briefings, seminars and webinars, where poverty may have been a dimension of the reporting but was not the main topic. Third, the majority of the awards demonstrating a

⁶⁶ GCRF 'Foundational Evaluation' (Itad, 2019).

⁶⁷ DP-level KII.

stronger focus on poverty were either at the beginning of generating results or were experiencing delays as a result of Covid-19 adaptations.⁶⁸

Nevertheless, findings also indicate that some awards are routinely monitoring poverty and using findings to adjust methodologies or activities undertaken. A range of awards built a poverty analysis into the research framework from inception, working with targeted communities to monitor or communicate results.⁶⁹ In particular, the evidence indicates that Hubs with an intersectional focus are more likely to integrate poverty analyses into their monitoring, reporting and dissemination of results – primarily through undertaking comprehensive ToC processes and defining multiple poverty-related terms such as 'wellbeing', 'inequality' or 'capabilities'.

3.3 EQ 3: How is social inclusion/equality, diversity and inclusion framed and understood within the GCRF ecosystem?

3.3.1 How the GCRF ecosystem frames and understands social inclusion/equality, diversity and inclusion (SI/EDI)

Summary finding: At DP level, the framing of SI/EDI and formal alignment with ODA compliance measures is fragmented, although Delivery and Strategic Plans in 2019 demonstrate that there is progress in the promotion of SI/EDI concerns across a handful of DPs. At the award level, while similar progress is evident, the majority of awards do not systematically demonstrate any formal written processes for addressing SI/EDI compliance guidance from DPs. The recognition of 'equality, diversity, and inclusion' terminology is also patchy in terms of its diffusion across award holders – although this is growing awareness of this in many awards.

The conceptual framing of SI/EDI issues across the DP and award levels are largely dependent on the framing of the award, the team composition and the unique specialities of team members, rather than being grounded in written ODA compliance guidance. Nevertheless, there is a handful of awards that are demonstrating promising practice with respect to how they are mainstreaming SI/EDI concerns across both award management processes and the award thematic content itself. At the DP level there is promising momentum on the framing of SI/EDI concerns, given the creation of several Delivery and Strategic Plans in 2019.⁷⁰ These materials give some insights about how SI is ideally incorporated and generally understood within the GCRF ecosystem. For instance, ESRC's 2019 Delivery Plan explicitly expresses the ambition to strengthen the focus on EDI, while the AHRC Delivery Plan 2019 brings up the opportunity to focus on SI considerations, i.e. disability, LGBTQ and marginalised communities.

Similarly, NERC's Delivery Plan states its intention to revise its Responsibility Framework – drawing upon UKRI's EDI strategy in doing so. The BBSRC Delivery Plan 2019 is relatively blind to SI/EDI issues but does acknowledge the intention to incorporate SI/EDI considerations in their research – this is understood in terms of disability, gender reassignment, marriage/civil partnership, race, religion or belief, sexual orientation and age. BA's 2019 Strategic Plan

⁶⁸ G2, G5, G19.

⁶⁹ G17, G33, G30, G18, G15.

⁷⁰ ESRC, AHRC, MRC, BA, BBSRC and NERC.

includes SI as part of its vision for 2022. Similarly, MRC highlighted their intention to enhance the EDI strategy as a long-term aim in the 2019 Delivery Plan.

These delivery, action and strategic plans have emerged after UKRI developed a strategy and action plan for EDI in 2018–19. Within UKRI, this was followed by a formal requirement for all calls to incorporate an equality impact statement. This requirement seems to be reflected in some of the newer calls, which include requirements for inclusive advisory committees to advise on diversity and equality issues. However, only five out of 15 funding calls systematically address SI as part of the selection criteria and the core rationale – the 'One Ocean Hub' being a prime example. Also, the guidance from 2019 for gender M&E, focused on Hubs, specifically requests awardees to provide information about how each award has proven to have an impact on EDI, although reflections and specific guidance concerning disability issues are often under-represented.

Other DPs also seem to have included SI in the framing of funding calls, but the extent to which this is included as selection criteria is lower. For instance, ESRC incorporates SI as such in three out of six funding calls reviewed. BA addresses SI in 11 of the 12 calls, and in four of them as selection criteria. Most of these show evidence an intersectional understanding of SI, including gender, age, disability, ethnicity, race, religion and spatial factors.

A range of other ad hoc actions supporting the mainstreaming of SI/EDI is observable at DP level. These include the elaboration of guidance for applicants and reviewers (BA, RAEng), programme theories of change (Innovate UK) or internal policies and statements that address SI in different forms (RAEng, BA, MRC). For instance, BA have started to take SI into consideration in the assessment guidelines of three particular programmes – referring to ethnicity, race, disability, spatial factors, religion and age factors – as a mandatory requirement for applicants. RAEng guidance documents for applicants and reviewers include a diversity and inclusion statement, and some of the programme calls incorporate specific questions on how these dimensions are to be included in projects. The most recent documents include explicit requirements to increase the participation and under-represented groups (e.g. Frontiers and Higher Education Partnerships in Sub-Saharan Africa (HEPSSA) programmes).

However, this does not seem to be generalised among all DPs, and the degree to which SI is addressed varies. In several cases, regardless of having developed the aforementioned action and delivery plans, there is no observable action in other sections of the literature from 2019 and 2020. No significant evidence has been found among the remainder of DPs regarding SI considerations (RE, RE, SFC, AMS, DEFNI, Innovate UK, MRC). In some of these cases the submitted strategies and proposals incorporate some of these dimensions, but there is no evidence of it having been requested by the DPs.

At the award level, despite some growing recognition of SI/EDI terminology and compliance concerns accelerating at the DP level, research teams are largely unaware of formal guidance on SI/EDI with respect to internal research systems as well as application of SI/EDI to research content or enquiries. The topic modelling analysis shows that SI/EDI terms are largely framed around gender, voice, social justice, diversity, equality, empowerment and participation. Similar to the topic modelling findings on gender, the topic matches drew most results on the topics of maternal pregnancy, peace and violence, and civil society. However, fewer than 1% of results demonstrated a consistent (>20 keyword matches) reflection of SI terms (29% for >5 keyword matches, with only one result for >40-word keyword match).⁷¹

⁷¹ See Annex 7 for a list of GESIP-related keywords used in the topic modelling.

The award analysis also shows that a crude distinction can be made between social scienceoriented award holders as opposed to the physical or natural sciences (as noted in section 3.1.1). For instance, despite entertaining a 'holistic vulnerability model', the PI from an environmental risk-oriented award stated that 'We haven't really thought about this [SE/EDI]'.⁷² In another award focused on environmental dynamics, award holders stated that while SI was not formally considered in research aims or methods, nor in project management and the formation of associated teams, they were aware of a broader sea-change regarding the recognition of SI themes.⁷³

Another pool of awards, the majority of which tread the line between having a social science and physical science orientation, demonstrates a degree of recognition of SI/EDI issues in either teaming arrangements or research content. However, this recognition is usually framed around the importance of a single social axis (caste, socioeconomic status, ethnicity) and not a holistic or intersectional approach, and is largely developed on the basis of researcher interests and academic dispositions rather than formal guidance. For instance, one award focusing on governance issues in South Asia noted in the later stages of the research that certain social groups (castes) were under-represented in participatory consultations and were later added to the cohort list.⁷⁴ Another award, focusing on young people, sought to recruit a very specific of young enumerators into teaming arrangements,⁷⁵ while others sought to frame SI/EDI awareness in terms of 'community participation'⁷⁶ or the inclusion of other marginalised groups (Indigenous communities).⁷⁷

A final body of awards – primarily those that have a substantive or focused lens on social science topics – demonstrate a more comprehensive recognition of a range of SI/EDI concerns in their award documents and feedback. However, these awards – like those mentioned above – have largely under-written their commitments to mainstreaming SI/EDI issues, using either their teams' pre-existing inclinations or the requirements of the subject matter as justification rather than formal guidance from respective DPs. These awards tend to focus on vulnerable populations first and foremost, and disaggregate target groups by 'religion, ethnicity and language', 'geographic and communicational' separation, horizontal inequality, 'multiple age groups', 'ethnicity, age and refugee status', and 'caste, ethnicity, race – intersecting with gender'.⁷⁸ There are also indications that awards associated with Hubs are able to invest more time and effort into disaggregating and presenting SI/EDI issues in a more granular way.

A systematic approach for reviewing potentially relevant SI/EDI dimensions in research strategies and systems vis-à-vis ODA compliance expectations is largely lacking. Only a small number of awards go further in appreciating the full implications of an intersectional approach in research design and implementation.⁷⁹ One award in particular – situated within a Hub – has demonstrated the capacity and interest to go so far as to pilot different metrics for assessing equality gaps from an intersectional perspective.⁸⁰ The pilot activity is also designed with scalability in mind and, as noted by the PI, 'what we are testing can reverberate in different spaces'. This award, along with a small number of others, is able to demonstrate more interaction with DP programme officer staff, who have been supportive in reflecting on such

- ⁷³ CS9, CS10.
- ⁷⁴ CS16, CS17.
- ⁷⁵ G34.
- ⁷⁶ GR8.
- ⁷⁷ G21.

⁸⁰ G4.

⁷² CS24.

⁷⁸ G28, G31, G17, G29, G15, GR2.

⁷⁹ G2, G3, G4, G5.

lessons, and facilitating ad hoc horizontal learning activities – although the energy for these interactions is driven largely at the award or hub level.

However, on the subject disability in particular, very few (if any) awards aspire to take an active or peripheral interest in disability issues – either in isolation or as part of a wider package of potential vulnerability or inequality issues. Three of the sampled awards (3 of 54) undertake an explicit focus on disability or disabilities, while only one award attested to altering their research strategy to accommodate an emerging understanding of the importance disabilities in framing their results.⁸¹

3.3.2 Expertise available to address social inclusion in the GCRF

Summary finding: The majority of DPs and award holders appreciate the importance of SI/EDI issues in developing team structures and thematic expertise, while acknowledging the role that unequal power relations have in affecting team and partner relations. Overall, there is a strong emphasis on early career professionals – particularly individuals originating from the 'Global South'. However, the evidence suggests that the role of ODA compliance guidance in contributing to this progress has been limited.

At the DP level, there is evidence of mechanisms to address SI/EDI only among a good number of partners. This includes the establishment of teams such as UKRI's EDI External Advisory Committee, Innovate UK's Energy and Equality, Diversity and Inclusion team, or MRC's network of employees designated as 'equality champions'. The 'equality and unconscious bias' trainings facilitated by MRC, for both managers and employees across the organisation, also include SI/EDI dimensions. MRC has developed an adjustment guide to raise awareness and inform staff about the responsibilities of managers towards employees with disabilities. BA has also developed an 'Equality of Treatment' policy, while RS has created a tool to track diversity data on ethnicity and disability among programme.

ESRC highlighted, in their '2019 Delivery Plan', the ambition to improve how diversity data is captured and held with respect to award holders. More broadly, UKRI has incorporated specific requirements to include inclusive advisory boards to advise on matters of diversity and equality within recent projects. Other than these, no more evidence has been found among the remaining partners. However, there is negligible documentation at the DP level which explores how these tools and policies have been applied in practice or how outcomes or promising practice have been observed as a result of their use.

At the award level there are also promising signs of relevant SI/EDI expertise being sought and developed – with a particular focus on early career professionals. The dominant framing of SI/EDI issues with respect to teaming arrangements and associated capacity building across a minority of awards is one that considers the needs and interests of early career researchers. This framing often intersects with gender and geography in the sense that award holders refer to proactively engaging 'female early career researchers situated in LMICs'.⁸² However, the remaining number of award holders could not provide detailed insights on how SI/EDI expertise is addressed in their awards.

Where award holders did provide relatively clear detail on their approaches to address SI/EDI issues in relation to expertise, they almost routinely stated that this agenda was driven

⁸¹ G14, G19, CS18-20, and latterly G25.

⁸² CS45, CS10, G23, G8.

internally and without prompts in relation to ODA compliance guidance. There are also clear signs of a growing awareness of explicit 'EDI' terminology across most award holders – although when pressed, the majority of award holders are unable to reflect on the detail of this expanding lexicon and the formal implications it has on their ways of working.

A minority of award holders do consciously address SI/EDI expertise concerns – although in fairly simplistic ways. For instance, SI/EDI expertise considerations do not normally stretch beyond considerations for early career researchers, the engagement of a diverse team of enumerators or, at best, the additional incorporation of diverse respondents (e.g. in the case of participatory action research or similar ethnographic-based methodologies).⁸³ In addition to this, there are only a small number of awards that demonstrably intend to use sophisticated frameworks to consider how an SI/EDI lens applies to team expertise and capacity building.⁸⁴ For instance, one of these awards undertook a partnership with a leading US-based university specialising in similar subject matter in order to create additional space for diversity-responsive internship programmes for students.⁸⁵ Another award sought to undertake a similar exercise, albeit with a series of African universities, and with an explicit understanding of promoting bi-directional capacity building (thereby undermining the common framing of capacity building activities flowing from the 'Global North' to the 'Global South').⁸⁶

3.3.3 Monitoring and dissemination of social inclusion data, findings and research outputs

Summary finding: There is limited evidence that guidelines relating to the reporting or dissemination of SI or EDI issues have been made available at both the DP and award level, although award teams nevertheless develop ad hoc MEL systems and intend to ensure that several multimedia and multilingual communications outputs will assist in addressing EDI concerns.

Several mechanisms to capture and monitor SI data and to appropriately communicate findings have been found in the DP-level documents. However, these are rather limited and found only among a limited number of partners. UKRI's annual diversity data releases seem to be the most systematic and common mechanism found to disseminate SI data within GCRF (ESRC, AHRC, MRC, NERC, BBSRC and STFC are shown to participate in this). In addition, a number of isolated resources to capture SI/EDI findings for different levels of the GCRF ecosystem have been found among several partners. For instance, AHRC engages with an independent annual EDI monitoring mechanism, and according to their 2019 Delivery Plan SI/EDI data findings are planned to be used to strengthen diversity within the institution. MRC also seems to publish diversity data regularly, but there is limited data on its consistency and quality. Innovate UK monitors SI at the programme level (i.e. competition analysis, workshops, outreach and communications).

Of particular interest are the UKRI evidence reviews which are committed to 'transforming research and innovation environments in order to ensure that they are safe, open and inclusive'.⁸⁷ These documents provide a UK-based and international-level focus on EDI in R&I.

⁸⁶ G4.

⁸³ G31, CS45, CS10, G23, G8.

⁸⁴ G2, G3, G4, G5.

⁸⁵ G3.

⁸⁷ See Guyan & Oloyede (2019) and Moody & Aldercotte (2019).

The former found that small proportion of sources 'discussed other characteristics, such as age, disability or socio-economic status, as the primary target of interventions', while the latter found that 'Other characteristics such as disability, religious inclusion or age were less likely to be the primary target of interventions'.⁸⁸ Neither document examines or discusses ODA compliance or the roles and functions of GCRF. At the award level in the BEIS tracker there is no mechanism for capturing, monitoring or reviewing SI/EDI metrics. This is surprising given that categories of analysis exist for gender, biodiversity, desertification and climate change. However, this may be understandable given the complexity required for understanding the multiple variables that might apply to a singular metric (e.g. class, caste, socioeconomic status, geography, ethnicity, etc.).

Nevertheless, there is a stand-alone marker on the topic of disability (see Box 7 for a discussion of the scoring criteria). A summary of the score rating is provided in Figure 5 and shows that approximately 25% of awards (n=2,925 between 2016 and 2020) have not been assessed to date. The proportion of awards allocated a 1-rating ('Significant') is approximately 1.6% of the overall sample, while a 2-rating ('Principal') constitutes a total of 0.5% of the overall sample.

Box 7. BEIS tracker guidance for capturing progress on disability focus

Activities are classified as being inclusive of persons with disabilities if:

they have a deliberate objective of ensuring that persons with disabilities are included, and able to share the benefits on an equal basis to persons without disabilities

or

they contribute to promote and ensure the full and equal enjoyment of all human rights and fundamental freedoms by all persons with disabilities, and promote respect for their inherent dignity in line with Art. 1 of the Convention on the Rights of Persons with Disabilities

or

they support the ratification, implementation and/or monitoring of the Convention on the Rights of Persons with Disabilities.

2 - principal

1 – significant

0 - assessed but not targeted

Blank – not assessed.

Finally, an analysis of the awards that been designated a 0-rating shows a very small margin of error (only two awards have been miscoded). Overall, while the scoring criteria appear to be broadly accurate, the criteria do not disaggregate between internal teaming arrangements (i.e. the representation of disabled persons in research activities) and research content (i.e. queries concerning disabled persons in research questions or methods).

⁸⁸ Ibid.





The award-level analysis also reveals a fragmented picture in terms of the systematic and comprehensive capture of SI/EDI-related MEL data – and there is clear evidence that the majority of award holders have not been required to provide EDI statements as yet. For instance, the majority of awards do not provide disaggregation of MEL data by social categories,⁸⁹ and note that 'there we no specific boxes to fill in for social exclusion'.⁹⁰

However, there is a small number of awards that have maintained a proactive focus on SI/EDI concerns using different MEL and communication modalities. These awards have undertaken a small number of horizontal learning activities with like-minded award holders, with little facilitation guidance from GCRF in doing so. One award incorporates a concern for power dynamics – 'including the effect of socio-cultural established roles' – in understanding the power dynamics across the award.⁹¹ Another award fed back that they are engaging a 'wide cadre of thematic specialists using participatory approaches' – but recognised that the capacity to explore this to the desired extent was constrained, and that specific guidance on DP expectations was unclear.⁹² Other awards 'intend to ensure that several multimedia and multilingual communications outputs will assist in addressing EDI concerns', while another 'continues to develop multiple non-hierarchical communications and monitoring mechanisms'.⁹³

Critically, across the awards, an overarching mapping of research activities using the topic modelling (data science techniques and algorithms) shows that LGBTQIA+ issues are notably under-represented in the data set. Very little evidence was obtained in relation to efforts to map or monitor LGBTQIA+ issues within teaming or capacity building activities or within the

⁸⁹ CS6, CS8, CS10, G33, G17, G21, G22, G32.

⁹⁰ G14.

⁹¹ GR5.

⁹² GR6.

⁹³ G2, G3.

content of research themes. In fact, less than a handful of awards address or discuss LGBTQIA+ issues centrally or peripherally in their research focus.

3.4 To what extent is GCRF responsive to GESIP issues with respect to Covid-19 adaptations?

Summary finding: GESIP-related factors were only one factor in determining an individual award's response to the challenges raised by Covid-19. Explicit general guidance, either by the GCRF or by DPs, with respect to the gender, social inclusion and poverty dimensions was not found or reported. Given the uncertainties and unpredictability of the pandemic, it is hardly surprising that written guidance has not been issued. Despite the absence of guidance, GESIP-related dialogue has taken place, albeit in ad hoc ways and largely driven by the award and its research focus.

The majority of awards have engaged in multiple conversations with both partners and DPs in mapping and addressing Covid-19 implications for ongoing research activities. DPs have reportedly been fairly active in facilitating no-cost extensions and providing logistical guidance. A fair number of awards declare that DPs have supported them in undertaking content adjustments, but award holders generally suggest that DPs could be more active in facilitating adaptive measures and guidance to awards which are critically affected by Covid-19 challenges.

There is no documented evidence that GCRF has influenced how DPs have reacted to Covid-19 in relation to the GESIP dimensions of the pandemic. Evidence across the DP Review and awards suggests that no formal guidance for tailoring research activities was introduced or facilitated by the GCRF in response to the implications of Covid-19. However, given the volatility of the situation, DPs and awards report that they are responding on a case-by-case basis.

Awards already embedding gender and SI in their research reported that a variety of actions has been undertaken to capture the differential impacts of Covid-19 and/or safeguard local researchers, enumerators and participants.⁹⁴ As noted by one key informant, 'Different measures have been taken to try to address this challenge'.⁹⁵ This sentiment is echoed among a number of award holders, who generally express that they have undertaken rapid learning exercises with other award holders and have undertaken reviews of emerging resources to revise their data collection plans. As a result, award holders have been ensuring that appropriate safeguarding measures are in place for research teams and respondents, while also adjusting their research objectives to cater for a changing context.

3.5 Interlinkages with other modules

Three broad interlinking finding categories have been observed across the GESIP evidence base that are pertinent to the concerns of the Management, Fairness and, to a lesser extent, Relevance modules.

The first interlinkage speaks mostly to the Management module, in that a small but vocal number of vanguard or innovative awards – almost exclusively having a GESIP focus – expressed dismay in relation to the structural constraints of the GCRF award management

⁹⁴ G31, G9, G2, G8, G26, G12.

⁹⁵ GR2.

system.⁹⁶ Several award holders noted that the purpose of their award was to reverse or circumnavigate alleged 'neo-colonial' or 'traditional' academic models by facilitating southern partners to engage as co-leads in research activities, to manage and distribute funds accordingly and to engage in research that sought to 'reverse the Eurocentric gaze' by engaging actors in non-ODA compliant countries.⁹⁷ These award holders noted that the inflexibility of the GCRF management structure was an inhibitor for their award activities and for innovative experimentation.

Second, there are noticeable connections across the GESIP and Fairness module, given various findings of the GESIP module in relation to appropriate to the seeking and fostering of appropriate technical expertise. The findings across gender, SI/EDI and poverty sub-modules all concur that that there is a strong and inherent interest among award holders to balance inequalities between 'Northern' and 'Southern' researchers, particularly early career researchers. These findings are consistent across the sub-modules, and also demonstrate a general lack of sophistication among award holders with respect to undertaking strategic and tailored approaches to ensure diverse and inclusive teaming arrangements.

Third, the issue of horizontal learning and diffusion is likely to resonate with the Management module. The relatively independent, opportunistic and organic nature of learning observed among award holders across the GESIP sub-modules suggests there is a broader management implication for knowledge management considerations across the GCRF. As it currently stands, award holders are mostly sharing approaches to improve the GESIP responsiveness of their awards based on ad hoc personal networks and engagements. There are, therefore, wider managerial concerns for GCRF to consider with respect to its role in facilitating and resourcing knowledge sharing and capturing more innovative experiences observed among DPs and award holders.

⁹⁶ G2, G3, G4, G5.

⁹⁷ CS16.

4 GESIP audit conclusions

This section provides an overview of the GESIP audit conclusions for each EQ and summarises findings in terms of the more promising results as well as the ongoing challenges across gender, poverty and SI concerns.

The aim of the GESIP audit is to provide an assessment of the extent to which GESIP concerns have been addressed throughout GCRF at all levels of operation between 2016 and 2020. The conclusions are of key concern to GCRF (and similar funds), DPs and award holders in that the effective mainstreaming of GESIP issues across the fund is not only a matter of accountability but also one of development impact. UK law (2010, 2014) requires that ODA spending addresses a range of potential social inequalities⁹⁸ and can demonstrably lead to 'poverty alleviation' in ODA contexts.

While this issue is partially a 'compliance' concern, it also stands to reason that development impacts – as a result of UK funding – are likely to be more comprehensive and sustainable if GCRF stakeholders engage in GESIP mainstreaming activities. Ideally these activities would seek to surpass 'compliance' expectations and pave the way for a more aspirational focus on 'excellence'.

The conclusions below demonstrate that the GCRF ecosystem, as it currently stands, is moving positively towards an improved compliance culture in relation to ODA expectations on GESIP. Shifts in this direction are more evident with respect to GE concerns and, to a lesser extent, across SI/EDI concerns. Compliance concerns regarding the framing and addressing of 'poverty' are shown to be the least fulfilled and fairly problematic. The reasons for this mixed performance across the GESIP areas are outlined below, and although there are pockets of promising or best practice across all GESIP areas, there is still some way to go before GCRF can foreseeably demonstrate a shift from a culture of ODA compliance towards a culture of excellence in addressing GESIP concerns.

The conclusions demonstrate that some of the recommendations of the Tetra Tech 'Review of Approaches to Gender Equality' report (2020) (see section 1, Box 2) and EDI reviews (2019) have been gaining momentum.⁹⁹ For example, the positive trajectory of growth leading towards comprehensive systems, policies, expertise and MEL systems for GESIP-related compliance is demonstrating opportunities for improvement and consolidation (Tetra Tech recommendations 1–3). Gaining traction on the fourth recommendation of the Tetra Tech report will also be particularly challenging, given challenges noted around the strategic development and maintenance of GESIP-related expertise, capacity building and monitoring systems.

⁹⁸ Age, disability, gender, relationship status, race, religion or belief, sex and sexual orientation.

⁹⁹ **Recommendation 1**: BEIS should develop a clear, coherent approach to gender equality, diversity and inclusion at the ODA portfolio level.

Recommendation 2: BEIS should institute a systematic approach to learning, clearly differentiating between internal and external functions at key stages in the fund management cycle.

Recommendation 3: BEIS should develop a fund level approach to gender equality, diversity and inclusion that is tailored to each of the funds.

Recommendation 4: BEIS should ensure there is dedicated senior management capacity, sufficient resourcing and clear accountabilities at all levels to design, implement, monitor and adapt the approach.

Overall, the conclusions show a number of ongoing gaps and challenges that suggest that GCRF could have been in a better place with respect to mainstreaming of GESIP concerns had more proactive attention been paid to fairly fundamental coordination, facilitation, monitoring and learning functions (i.e. the 'broken trails'). The metaphor of a 'broken trail' is used frequently to identify where specific attention is required. This term has variable meanings in the report, but 'broken trails' can refer to the unfulfilment of accountability measures being in place or acted upon (conclusions 1 and 3), the limited capture of quality MEL compliance data (conclusions 2 and 3) and gaps in compliance guidance between higher-level ODA commitments and award holders (conclusions 2 and 3). This 'broken trail' is particularly notable at the interface between DPs and award holders and in the limited dialogue and accountability measures enacted beyond the proposal stage.

4.1 Gender equality: conclusion, improvement opportunities, and ongoing gaps and challenges

Conclusion 1: The mainstreaming of GE concerns across GCRF is moving at a gradual pace and is being supported by a variety of mechanisms from the DP level through to awards – although strategic steers at both levels are mostly uncoordinated, leading to a 'broken trail' of communication and accountability between these two levels of activity. At the DP level, there is strong evidence showing that gender mainstreaming at the level of strategies, policies and processes has been taking a foothold since 2018/2019, and that there is a gradual but ad hoc uptake of policies and understanding concerning genderresponsive expertise. As a result, while progress is gradual, GCRF is broadly 'on track' at the DP level in terms of mainstreaming gender compliance concerns – but could move in a faster and more coordinated way while also facilitating horizontal learning, with the added benefit of celebrating 'excellence' as much as 'compliance' results with award holders.

At the award level, however – while there are discrete and promising pockets of activity with respect to developing systems and creating mechanisms for MEL – the overall gains are fragmented, uncoordinated and largely driven by award holders themselves. In addition, moments for award holders to engage in formal accountability procedures are few, and are largely concentrated at the proposal and panel review stage. As a result, significant attention is required at the award level to align with progress at the DP level.

4.1.1 Promising results

Several promising results exist regarding the institutionalisation of gender mainstreaming in the strategies and systems of DPs and award holders. There is clearly evidence of traction in this area at the DP level as of 2018 and 2019, when UKRI (unilaterally) introduced Gender Equality Statements and the Gender Equality Act Compliance Procedure. The development of a Gender Equality Toolkit by Edinburgh University is a particularly promising venture that can be either scaled outwards or carefully duplicated to ensure common gender mainstreaming denominators. There are signs that this process is already under way, in the form of informal cascading and collecting learning processes at the DP and award level – but these approaches are largely dependent on infrequently available spaces for sharing as well as on the inclinations of energised groups of individuals.

There are also several positive results regarding the interpretation of compliance guidance with respect to the mainstreaming of gender-responsive expertise across the GCRF system. At the award level there exists a strong predisposition for teams to understand and appreciate

the importance of gender diversity in teaming arrangements. While there is also a fairly simplistic approach focused on 'team parity' in most awards, a small number of awards are moving towards excellence in seeking to understand the highly variable gender-specific barriers and enablers that some individuals face in the GCRF system. For example, some research Hubs have proved able to access 'gender champions' as well as mandated gender experts in Executive Advisory Boards.

For MEL systems, there exist pockets of horizontal learning which are shown to be the dominant mechanism for diffusion. While promising MEL guidance exists (UKSA, UKRI), and while there is good momentum to take forward the onboarding of the Research ODA (RODA) monitoring tools across DPs, it is also recognised that 'no one size fits all' (UK parliamentary response to the Newton Fund Review, 2020). The ongoing use of the BEIS GE marker is another promising practice – although some issues remain with respect to data capture and the lack of differentiation between gender-responsive teaming arrangements and research content/methods. In terms of accountability, there is a positive story in relation to the maintenance and improvement of gender-responsive reflections at the proposal and panel review stage, where DPs and award holders intersect most systematically.

4.1.2 Ongoing gaps and challenges

The ongoing gaps in relation to the development and preservation of strategies, policies and systems for promoting GE can be located at both the DP and award level. Both levels of activity demonstrate a notable degree of fragmentation and lack of coordination in terms of understanding and addressing gender-related compliance concerns, leading to significant variability in the way gender mainstreaming strategies are conceived and promoted. At the award level, familiarity on expectations for gender mainstreaming in relation to ODA compliance remain embryonic and, as a result, award holders use their own initiative and framing in developing relevant strategies and systems. The form and function of these approaches are, for the most part, dependent on the predisposed interests and experience of award teams rather than on formal or informal guidance from DPs.

A number of key coordination gaps for promoting and retaining appropriate gender expertise across the GCRF ecosystem can also be located at the DP and award level, but these are primarily evident at the award level. Award holders are largely aware of the need for obtaining team parity, but this awareness is almost exclusively generated internally among award holders, particularly among awards focused on the social sciences and humanities. In addition, very few awards demonstrate a sophisticated capacity building approach in growing gender expertise in practice, while fewer still capture such approaches in formal commitments.

With respect to gender-responsive MEL systems at the DP level, the main ongoing gap relates to a lack of systematic coordination and the facilitation of shared learning spaces. At the award level, the main gaps are the limited familiarity of ODA expectations regarding reporting and an often idiosyncratic and simplistic understanding of the nature of gender-responsive MEL. These findings echo those recently presented in a 'Review of Approaches to Gender Equality' (Tetra Tech, 2020) in that 'internal organisational initiatives' tend to drive commitments to GE, while 'comprehensive monitoring frameworks or implementation plans' are generally underdeveloped. Overall, well-resourced knowledge management roles are largely absent across DPs and award holders.

There is also a lack of formal accountability procedures in relation to ODA compliance on GE commitments beyond the proposal and panel review stage: the majority of award holders are unfamiliar with the need to develop Gender Equality Statements and are rarely engaged by DPs on their performance if they do send gender-related monitoring metrics up the

management chain. Given the constraints in these systems, it is also understandable that gaps exist with respect to the gender-responsive framing of communication and dissemination approaches. Efforts in this regard are confined to 'including women and girls' in consultation processes and audience profiles, with little to no strategic or tactical awareness of when, why and how to mainstream gender-responsive elements in communications and outputs.

4.2 Poverty alleviation: conclusion, improvement opportunities, and ongoing gaps and challenges

Conclusion 2: There is limited evidence of a coherent framing for poverty or 'poverty alleviation' guiding GCRF, its DPs and award holders. Under the umbrella of the SDGs, poverty is implicitly and simplistically assumed to be core to all R&I activities undertaken through GCRF, although there is a wide range of framing and understanding of poverty-related terminology across the system. As a result, there is continuing evidence of a 'broken trail' in the capture and dissemination of the poverty dimensions of GCRF-supported research alongside a strong indication of under-reporting of poverty results and outcomes. Consequently, there is a considerable opportunity for GCRF to revamp and energise activities at all levels to understand multiple definitions of poverty and to facilitate all GCRF stakeholders to move beyond a simplistic presumption of automatic ODA compliance towards a framing that facilitates collective learning. The latter approach requires a platform and facilitation space for DPs and award holders to share their understanding of how research eventually contributes to poverty alleviation, and to frame this contribution into a broader narrative that GCRF will be accountable for.

4.2.1 Promising results

There is some fragmented evidence to suggest that some DPs have taken steps to define expectations on the framing and understanding of poverty in funding calls (UKRI and ESRC), delivery plans (ESRC, RAEng) or as part of communication material (RAEng, UKSA). The Gender Equality Toolkit, developed at the University of Edinburgh, also provided explicit guidance on the connections between poverty and gender inequality. In terms of developing appropriate MEL systems to capture GCRF contributions to poverty, the upcoming redesign of the RODA system is promising. The evidence also shows that a considerable number of workshops, briefings, seminars and webinars are being undertaken among DPs and award holders that may offer shared insights on how poverty is being addressed.

4.2.2 Ongoing gaps and challenges

There is limited detailed guidance at the DP level or award level for understanding how poverty is framed and understood in GCRF, and associated ODA compliance expectations are fairly simplistic. This is partly the result of the International Development Acts (2002, 2014) not including any requirements in this regard. As a result, the interpretation of the framing of poverty has been left open to individual award teams, many of whom understand their research to be automatically relevant to poverty alleviation, given that their activities are being undertaken in ODA-listed countries (specific LMICs). Furthermore, topic modelling evidence points towards a potential bias in the framing of poverty in terms of econometric or income-based categories, although feedback from respondents generally revealed a more complex picture in which terms such as 'vulnerability', 'well-being', 'risk' and 'inequality' were routinely featured. There are also limited expectations regarding the development and capacity building of poverty-specific expertise across DPs and award holders. Both the documentation and the feedback from key respondents suggest that understandings of poverty expertise are largely assumed as implicit, given the nature of award holder experience and the context in which awards are being implemented.

With respect to MEL systems that cater explicitly for the framing of poverty in GCRF, the evidence reveals a problematic picture. Given the variability afforded to DPs and award holders in framing and understanding their approaches to poverty, it follows that associated MEL metrics and contribution stories are highly variable. No evidence in the document analysis or from DP or award-level key informants indicated a coherent system or mechanism to either unpack the concept of poverty or to register the pathways to impact in relation to award objectives. As a result, the ability to map how GCRF is contributing to pro-poor outcomes in ODA contexts is severely constrained. Without any systematic knowledge management oversight from GCRF or DPs, learning and elevating findings into a broader contribution story on poverty alleviation (which could include interlinkages between gender and SI/EDI) are constrained.

4.3 Social inclusion/equality, diversity and inclusion: conclusion, improvement opportunities, and ongoing gaps and challenges

Conclusion 3: The mainstreaming of SI/EDI compliance concerns is fragmented at both the DP and award holder level, although there are promising signs of downward diffusion of EDI language and concepts in a good number of DPs and awards. Nevertheless, most award holders are not familiar with formal ODA compliance in relation to SI/EDI, and the majority of efforts and thinking is dependent on the idiosyncrasies of the award subject matter and the unique specialities or interests of award team members. As a result, and in line with preceding conclusions, there is evidence of a 'broken trail' of managerial guidance and accountability with respect to the mainstreaming of SI/EDI issues across GCRF.

Nevertheless, awards largely acknowledge the role that unequal power relations have in affecting team and partner relations. This awareness is often simplistically couched in terms of support for early career researchers in the 'Global South' rather than in any comprehensive strategic approach, which is in turn not informed by any overarching guidance from DPs. Finally, systems for framing and reviewing disability concerns – and, to a much lesser extent, LGBTQIA+ concerns – are notable among only a small handful of DP and award holders. Consequently, progress on the mainstreaming of SI/EDI compliance concerns is somewhat behind that of gender compliance mainstreaming concerns – particularly at the award level – while a small number of DPs and awards are actively endeavouring to innovate for 'excellence', rather than 'compliance', in SI/EDI mainstreaming.

4.3.1 Promising results

Despite some shortfalls on the promotion of strategies, policies and processes for ensuring ODA compliance on SI/EDI issues, there is, nonetheless, good momentum at the DP level in this regard. The Delivery and Strategic Plans of UKRI offer a solid foundation through which to continue to embed strategic approaches to for SI/EDI mainstreaming, while a number of ad hoc activities among other DPs – RAEng, BA, MRC and Innovate UK in particular – give good

examples of organisational ToCs or other internal policies and statements to address top-level framing of how to promote SI/EDI.

As stated above, while this momentum is still beginning to formally trickle down at the award level – demonstrated in the growing familiarity of 'EDI' terminology across most award holders – there are pockets of activity that can be reviewed and amplified for more systemic learning and knowledge management purposes. For this purpose, a rough distinction can be made between social science or humanities-oriented awards as opposed to the physical or applied sciences, where the former have a noticeably greater tendency to address SI/EDI concerns in the strategic framing of awards. However, even only a small number of social science awards move beyond the application of a single social axis lens (e.g. caste, class, ethnicity or disability) in research content. As a consequence, a small cadre of awards does present a comprehensive intersectional approach in their research ambitions and activities and are thereby demonstrating 'excellence' rather than 'compliance' in terms of ODA compliance.

There are also a few relatively advanced award holders that take a comprehensive 'bidirectional' learning and capacity building approach with respect to building relevant SI/EDI expertise. While such awards are few, they can be seen to be at the vanguard of potential learning opportunities to address SI/EDI issues in teaming arrangements and associated support structures, as these award holders conceive of SI/EDI compliance beyond addressing early career researchers per se. At the DP level, UKRI's EDI External Advisory Committee, Innovate UK's Energy and Equality, Diversity and Inclusion team and MRC's network of employees designated as 'equality champions' are particularly promising.

With respect to SI/EDI-sensitive MEL systems at the DP level, UKRI's annual diversity review is promising in that this provides a common mechanism to map and disseminate SI/EDI data in GCRF (ESRC, AHRC, MRC NERC, BBSRC and STFC are known participants). Other positive findings are noted with respect to AHRC, MRC and Innovate UK, each of which engage with their own independent annual EDI monitoring mechanisms.

4.3.2 Ongoing gaps and challenges

While there are promising results emerging at the DP level since 2018/19 with regard to the mainstreaming of SI/EDI issues across GCRF, these efforts lack guidance at a challenge level and are not yet coherent across DPs, leading to the development of a divergent range of strategies, policies and systems for ensuring ODA compliance on SI/EDI. Guidance is not only fragmented at the DP level, but actions to drive more active engagement from awards have been limited, resulting in award holders often applying their own framing and understanding of SI/EDI compliance criteria – almost always in the proposal and/or panel review processes. This has meant that the framing of SI/EDI at award level has largely been driven by the nature of the award in question and by the associated team as the awards continue into the implementation stages.

There are also gaps in the framing of polices and systems from both DPs and awards to address potential discrimination on the grounds of disability and gender identity (LGBTQIA+) as well as in guidance to promote these social dimensions in appropriate research content. Guidance on the framing of gender identity dimensions is particularly absent across GCRF in this regard.

In terms of MEL systems, there is limited evidence that guidelines from DPs relating to the reporting or dissemination of SI/EDI issues have been made systematically available to award holders beyond the ad hoc dispensation of requests for EDI statements. In addition, at the award level there is no tracking mechanism for capturing or compressing SI/EDI monitoring data into indices or a single index. In particular, the lack of a mechanism for capturing the

depth and breadth of uptake on LGBTQIA+ concerns in teaming arrangements or research content represents a gap for MEL systems between the DPs and award holders. Finally, it should be noted that the scope and reach of the disability marker in the BEIS tracker has overlooked up to 25% awards for assessment purposes and, like the gender marker in the BEIS tracker, it does not address concerns for both disability-responsive teams and research content.

4.4 GESIP-sensitive Covid-19 responses: conclusion, improvement opportunities, and ongoing gaps and challenges

Conclusion 4: GESIP was only one factor in determining an individual award's response to the challenges faced because of Covid-19. Explicit general guidance, either by the GCRF or by DPs, with respect to the gender, social inclusion and poverty dimensions was not found or reported. Given the uncertainties and unpredictability of the pandemic, it is hardly surprising that written guidance has not been issued. The absence of guidance does not mean that dialogue and more formal guidance between some award holders and some DPs or Hubs is non-existent: evidence from the award holder review suggests that GESIP-related dialogue was prominent, but largely driven by the interests of award holders (primarily in relation to methodological adaptations).

4.4.1 Promising results

DPs and awards are demonstrating some adaptive management lessons in responding to Covid-19, although these potentially innovative research methods and approaches have been developed primarily among award holders. Importantly, a Covid-19 'lens' thereby introduces a space where GCRF, DPs and award holders can begin to better understand the interlinkages between gender, poverty and SI/EDI. In effect, this constitutes an opportunity to consolidate cross-cutting lessons across GCRF in the form of a series of Covid-19 case studies that can be made accessible and/or tailored for groups of award holders.

4.4.2 Ongoing gaps and challenges

Given the novelty of the Covid-19 experience, it is difficult to determine the roles and responsibilities of the GCRF in relation to DPs and award holders with respect to Covid-19 adaptation processes. Nevertheless, there is good evidence to suggest that there has been little concerted effort across the GCRF, DPs and awards to synthesise lessons around Covid-19 adaptation measures. Award holders have noted that DPs could have been more proactive in filtering relevant information, or more aware of the need to facilitate dialogue between awards that might be well placed to share Covid-19-specific learning with each other. Rather, the focus of GCRF and DPs has reportedly been managerial in nature, in that support has largely been oriented to costed and non-costed extension discussions.

5 Recommendations

This section sets out recommendations designed to support GCRF move from a culture of ODA compliance towards a culture of excellence in addressing GESIP concerns. The four interlinked recommendations work to consolidate and strengthen the progress made to date to mainstream GE and address systematically the SI and poverty dimensions of ODAsupported research.

They build on the review of GE approaches in the GCRF, its guiding principles and its ToC. The recommendations are framed to inform the development of the GCRF's Gender, Equality, Diversity and Inclusion Strategy and optimise interlinkages between gender, SI and poverty alleviation – the overarching goal of the SDGs.

5.1 Overview

Standardising the processes and practices for mainstreaming GEDI and addressing the poverty dimensions of research will support GCRF to demonstrate tangible progress towards its intended outcomes and impact. Progress since 2019, while inconsistent, has been relatively rapid. Now is the time to consolidate and amplify progress.

The GESIP audit's recommendations are made in the light of the changing context of ODA research and the proactive positive response of GCRF to the 'Review of Approaches to Gender Equality' (Tetra Tech, 2020). Actions to strengthen GCRF's approach to gender and SI include the setting up of a working group to develop a gender KPI, the further appointment of a senior social researcher to BEIS and the ongoing development of the GEDI strategy.

The recommendations are therefore framed to strengthen the interlinkages between poverty alleviation, gender and SI identified in the findings and conclusions of the audit. In doing so, these recommendations also build on the GCRF's key principles which, while inconsistently integrating GESIP, provide the overarching strategic direction towards ODA excellence and its ToC, which embeds a pro-poor focus, gender and SI dimension within it. Both the principles and the ToC are fundamental building blocks for designing and implementing a GESIP strategy grounded in the vision and operating context of the GCRF.

The recommendations were informed by discussions with the participants of the two learning workshops, held in January 2021. These workshops enabled the GESIP team to ground truth the findings and conclusions as well as to identify realistic and doable actions for enhancing GCRF's capacity to influence the integration of GESIP issues across its ecosystem throughout the funding cycle.

5.2 Recommendations

Recommendation 1: Embed GESIP in the GCRF principles

The ongoing development of the GEDI strategy offers the opportunity to review the narrative of GCRF's key principles (Challenge and Impact Focus, Interdisciplinary Research

Excellence, Global Partnerships, Organisation and Leadership) in relation to the objective of making explicit the GESIP-related compliance expectations of the fund.

With respect to reframing the narrative of each of the four key principles, discussions can focus upon:

- **Challenge and Impact Focus:** Co-creation with DPs, challenge leaders and Hubs drawing on the lessons learned within GCRF to develop the new RODA system, enhancing the BEIS tracker tool, and a focus on award holder experiences in developing more nuanced GESIP impact measurement approaches particularly in light of the withdrawal of Pathway to Impact Statements.
- Interdisciplinary Research Excellence: At the DP and award level, recognise and cater for the different situations and interests of humanities/social science disciplines as compared to physical/natural sciences with respect to the degree of experience and interest in mainstreaming GESIP expectations.
- **Global Partnerships:** At the award level, seek to shift the predominant narrative beyond that of 'team parity' and a focus on early career researchers towards a more strategic approach that recognises wider barriers and enablers for GESIP-responsive partnerships.
- **Organisation and Leadership:** In consultation with a spectrum of DPs and award holders, create a well communicated and easily accessible action plan for rolling out the new narrative across the GCRF ecosystem. The action plan should set timebound milestones and an end date for roll-out. See also Recommendation 4 (Open up learning spaces for GESIP that is facilitative and reflexive).

The process of reframing the key principles narrative should underpin and run in parallel with the development of the GEDI strategy. This will optimise cross-fertilisation between the two activities and build ownership across the whole GCRF ecosystem for the GEDI strategy.

Recommendation 2: Integrate a clear poverty perspective into the GEDI strategy

Poverty compliance guidance constitutes the weakest part of the GESIP compliance infrastructure across GCRF. By further clarifying GCRF's expectations and framing in relation to the interlinkages between poverty on the one hand and GE and SI/EDI on the other, a broader poverty impact narrative can be developed over time. This foundation provides an avenue for GCRF to assess both compliance and excellence.

While a critical interface will be at the call and in the proposal stage, clarity is needed as to how these interlinkages apply at all stages of the funding management cycle. In support of the ongoing development of the GEDI strategy, it is recommended that BEIS work with the fund to:

- identify poverty expertise from within the DPs and award holders (as well as externally) and set up a reference group to advise and inform the development of a poverty-sensitive strategy – while drawing on existing promising practice (see Edinburgh University Toolkit)
- develop a comprehensive pro-poor GEDI ToC nested under the overarching ToC, drawing on lessons from DPs' use of organisational ToCs. From this, produce an action and resourcing plan with monitorable milestones for rolling out and socialising the strategy
- nuance and update the definition of 'poverty alleviation' in a pro-poor GEDI ToC, thereby allowing researchers to invest in their own terminology and approach within those parameters

 develop a monitoring framework for both learning and accountability that allows for the mapping of different pathways that would viably lead to 'poverty alleviation' (or clearly related concepts – such as 'well-being' or 'empowerment' – and which recognise the contributions of GE and SI/EDI in these pathways).

Recommendation 3: Standardise the MEL infrastructure and communication of GESIP expectations (and associated resource support) across GCRF and its funding cycles

Fundamental to consolidating and expanding the emerging and promising practices among some DPs in the way GESIP expectations are communicated is the standardisation of these expectations throughout the GCRF system. It is recommended that GCRF:

- strengthen and standardise activities at the pre-proposal stage designed to support
 researchers understand what both compliance and excellence look like in relation to
 GESIP mainstreaming. Actions can include fixed-rotation workshops around
 expectations and setting up a repository (a centrally procured and managed clearing
 house) for GESIP guidance materials (toolkits, recorded webinars, etc.)
- review the time frame of each funding cycle and extend the time between the call for proposals and the application deadline, to enable thoughtful and nuanced engagement with researchers on the approach and design of the proposal with respect to GESIP concerns
- extend the use of gender champions to include SI/EDI and poverty champions across the whole ecosystem, and introduce dedicated poverty and SI/EDI expertise into the review process and selection panels
- facilitate access to GESIP expertise and knowledge repositories after a proposal has been accepted during and after an inception period for funded research
- embed a GESIP focus in the redesign of the RODA system and strengthen associated reporting capacity – particularly at the award level. GCRF could also work with DPs to increase the accuracy, visibility and – where possible –complexity of disability and LGBTQIA+ data categories in the BEIS tracker and ResearchFish platform
- explore ways of encouraging DPs to develop harmonised formats for identifying and tracking pathways to impact that explicitly identify how the research will contribute to the 'achievement of the SDGs, enhancing people's wellbeing, improving equality for people of all genders, promoting social inclusion, economic development and environmental sustainability in developing countries' (GCRF ToC).

Recommendation 4: Open up learning spaces for GESIP that is facilitative and reflexive as part of an increased emphasis on GESIP-oriented knowledge management for both compliance and excellence

Building on observed and growing communities of practice and horizontal learning mechanisms, GCRF can work with DPs and award holders to identify entry points and actions for developing a GESIP focus within GCRF's knowledge management system. This recommendation supports recommendation 2 of the Tetra Tech (2020) 'Approaches to Gender Equality Review' – 'BEIS should institute a systematic approach to learning, clearly differentiating between internal and external functions at key stages in the fund management cycle'. A more GESIP-focused knowledge management system will serve to better coordinate actions for capturing and disseminating GESIP-specific lessons on compliance and excellence.

To build this system, GCRF should:

- consider the development of a specific GESIP knowledge management strategy that is referenced in the upcoming GCRF GEDI strategy that standardises the varied stocks and flows of GESIP-related compliance approaches across DPs and award holders
- sensitise knowledge managers under an emerging overarching learning approach to the specific GESIP-related information needs and requirements of different DPs and award holders
- institute a reflexive and reflective learning approach for GESIP by facilitating a process
 of bringing together cohorts of projects both GESIP-focused and non-GESIP-focused
 to think through challenges and issues relating to mainstreaming issues in both
 research management and research content
- extend and celebrate the use of case studies from DPs and award holders that capture GESIP-related lessons and practices across the GCRF ecosystem particularly those that focus on excellence rather than compliance
- work more closely with DPs and award holders to create and facilitate both online and offline spaces that build on peer-led horizontal learning exchanges already in existence.

References

General/External

Academy of Medical Sciences, 2018. Award Terms and Conditions. Accessed February 5, 2021. https://acmedsci.ac.uk/file-download/16363745. Academy of Medical Sciences, n.d. Global Challenges Research Fund Networking Grants. Accessed February 4, 2021. https://acmedsci.ac.uk/grants-and-schemes/grant-schemes/gcrf-networking-grants. Academy of Medical Sciences, n.d. Guidance notes for completing the online application form. Accessed February 5, 2021. https://acmedsci.ac.uk/file-download/38678-57b56f8d48921.pdf. Academy of Medical Sciences, n.d. The Academy of Medical Sciences GCRF Networking Grant Conditions. Accessed February 5, 2021. https://acmedsci.ac.uk/file-download/91277119. Ahmed, Bayes, Md Shahinoor Rahman, Peter Sammonds, Rahenul Islam, and Kabir Uddin, 2020. Application of Geospatial Technologies in Developing a Dynamic Landslide Early Warning System in a Humanitarian Context: The Rohingya Refugee Crisis in Cox's Bazar, Bangladesh. Geomatics, Natural Hazards and Risk 11, no. 1: 446–68. Accessed February 25, 2021. https://doi.org/10.1080/19475705.2020.1730988. Beall, J. and Piron, L., 2005. DFID Social Exclusion Review. ODI. Accessed February 5, 2021. https://www.odi.org/sites/odi.org.uk/files/odi-assets/publications-opinion-files/2301.pdf. Bebbington, Jan, and Jeffrey Unerman, 2018. Achieving the United Nations Sustainable Development Goals: An Enabling Role for Accounting Research. Accounting, Auditing & Accountability Journal 31, no. 1: 2–24. Accessed February 25, 2021. https://doi.org/10.1108/AAAJ-05-2017-2929. Bradley, Richard, Serge Guillas, Garima Jain, Cassidy Johnson, Teja Malladi, and Julia Wesely, 2019. Integrating Tsunami Risk Assessments in Development Planning: Lessons from Western India. UCL Open: Environment Preprint. Accessed February 25, 2021. https://doi.org/10.14324/111.444/000026.v1. British Academy, 2017. British Academy Early Childhood Development List of Awards. Accessed February 3, 2021. https://www.thebritishacademy.ac.uk/documents/100/2017-09-05 British Academy Early Childhood Development - List of Awards.pdf. British Academy, n.d. Economic Conditions in Early Childhood and the Inter-Generational Transmission of Poverty. Accessed February 3, 2021. https://www.thebritishacademy.ac.uk/projects/early-childhoodeconomic-conditions-inter-generational-transmission-poverty/. British Academy, 2019. GCRF Networking Grants. Accessed February 5, 2021. https://www.thebritishacademy.ac.uk/programmes/gcrf-networking-grants/. British Academy, n.d. Towards Inclusive Health Systems and Infrastructure Access: Enhancing the Well-Being of Refugees in East African Cities. Accessed February 3, 2021. https://www.thebritishacademy.ac.uk/projects/cities-towards-inclusive-health-systems-infrastructureaccess/. British Academy, n.d. Understanding the Dynamics of Water Scarcity and Violence in Kenya. Accessed February 25, 2021. https://www.thebritishacademy.ac.uk/projects/heritage-dignity-violenceunderstanding-dynamics-water-scarcity-and-violence-kenya/. British Academy, 2019. Youth Futures Scheme Notes. Accessed February 4, 2021. https://www.thebritishacademy.ac.uk/documents/1817/Youth-Futures-scheme-notes.pdf. Cavaghan, Rosalind, n.d. Gender, Science, Technology and Development: Review for the Edinburgh University's Working Group on Gender in International Development Research. Accessed February 24, 2021. Chautard, A and Hann, C., 2019. Developing Inclusive Conferences. Accessed February 24, 2021. Choudhury, Tanzina, Sumena Sultana, and Suzanne Clisby, 2020. Conjugal Relations and Stepchildren's Wellbeing: Exploring the Experiences of Remarried Women in Bangladesh. Journal of Family Issues 41,

Climate Alliance Mapping Project, n.d. Accessed February 3, 2021. <u>https://climatealliancemap.org/</u>. Cochrane Library, 2019. Assessing the Appropriate Dose and Efficacy of Bovine Lactoferrin to Correct Iron Deficiency Anaemia in Pregnancy: A Community-Based Randomized Controlled Trial in Mirpur, Dhaka. Accessed February 25, 2021. <u>https://doi.org/10.1002/central/CN-01892150</u>.

no. 7: 1033–54. Accessed February 25, 2021. https://doi.org/10.1177/0192513X20918440.

Crafts Abilities, 2003-2021. Accessed February 3, 2021. https://www.craftsabilities.org/.

Devex, 2020. Poverty Reduction Missing from New UK Aid Strategy. Accessed February 25, 2021. https://www.devex.com/news/sponsored/poverty-reduction-missing-from-new-uk-aid-strategy-98655.

- Devex, 2020. UK Aid to Refocus on Countries Where 'interests Align. Accessed February 24, 2021. https://www.devex.com/news/uk-aid-to-refocus-on-countries-where-interests-align-98648.
- Earth and Space Science Open Archive, 2020. Probabilistic Quantification of Tsunami Currents in Karachi Port, Makran Subduction Zone, Using Statistical Emulation. Accessed February 4, 2021. https://www.essoar.org/doi/10.1002/essoar.10502534.1.

Europe PMC, n.d. Building Resilience and Resources to Overcome Depression and Anxiety in Young People from Urban Neighbourhoods in Latin America. Accessed February 3, 2021. <u>https://europepmc.org/grantfinder/grantdetails?query=pi%3A%22Priebe%2BS%22%2Bgid%3A%22MR%</u> <u>2FS03580X%2F1%22%2Bga%3A%22Medical%20Research%20Council%22</u>.

Foreign, Commonwealth & Development Office, 2020. *Letter from the Foreign Secretary regarding the future of the UK aid budget*. Accessed February 5, 2021.

https://committees.parliament.uk/publications/3683/documents/38142/default/.

- Gill, Andréa, and Thula Pires, 2019. From Binary to Intersectional to Imbricated Approaches: Gender in a Decolonial and Diasporic Perspective. Contexto Internacional 41, no. 2: 275–302. Accessed February 25, 2021. <u>https://doi.org/10.1590/s0102-8529.2019410200003</u>.
- GlobalGrace. Accessed February 3, 2021. https://www.globalgrace.net.
- GOV.UK, 2019. *Global Challenges Research Fund (GCRF): Foundation Stage Evaluation*. Accessed February 5, 2021. <u>https://www.gov.uk/government/publications/global-challenges-research-fund-gcrf-foundation-stage-evaluation</u>.
- GOV.UK, 2018. National Security Capability Review. Accessed February 5, 2021. <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/70</u> <u>5347/6.4391_CO_National-Security-Review_web.pdf</u>.
- GOV.UK, 2020. PM Outlines New Review to Define Britain's Place in the World. Accessed February 5, 2021. https://www.gov.uk/government/news/pm-outlines-new-review-to-define-britains-place-in-the-world.
- GRRIP, n.d. *GRRIP Project Grounding RRI Practices in Research Performing Organisations*. Accessed February 4, 2021. <u>https://grrip.eu/</u>.
- Guyan, Dr Kevin, Freya Douglas Oloyede, and Advance He, n.d. *Equality, Diversity and Inclusion in Research* and Innovation: UK Review. Accessed February 5, 2021. <u>https://www.ukri.org/wp-</u> <u>content/uploads/2020/10/UKRI-020920-EDI-EvidenceReviewUK.pdf</u>.
- Heidarzadeh, Mohammad, Alexander Rabinovich, Satoshi Kusumoto, and C P Rajendran, 2020. *Field Surveys and Numerical Modelling of the 2004 December 26 Indian Ocean Tsunami in the Area of Mumbai, West Coast of India.* Geophysical Journal International 222, no. 3: 1952–64. Accessed February 25, 2021. <u>https://doi.org/10.1093/gji/ggaa277</u>.
- IEED, 2019. Towards More Inclusive Urban Health Systems for Refugee Wellbeing. Accessed February 24, 2021. https://www.iied.org/towards-more-inclusive-urban-health-systems-for-refugee-wellbeing.
- IEED Publications Library, 2019. *Cities for All? Rethinking Urban Displacement*. Accessed February 3, 2021. <u>https://pubs.iied.org/17642IIED</u>.
- IIED Publications Library, 2019. *Dismantling Barriers to Health and Wellbeing for Nairobi's Refugees*. Accessed February 3, 2021. <u>https://pubs.iied.org/17714IIED</u>.
- IIED Publications Library, 2019. *Refugees in Cities: Grassroots Researchers Shed Light on Basic Needs*. Accessed February 3, 2021. <u>https://pubs-live.ac.iied.org/17643IIED</u>.
- IIED Publications Library, 2019. Towards More Inclusive Urban Health Systems for Refugee Wellbeing: Lessons from Kampala, Uganda. Accessed February 3, 2021. <u>https://pubs.iied.org/10881IIED</u>.
- IEED Publications Library, 2019. Urban Refugees in Nairobi: Tackling Barriers to Accessing Housing, Services and Infrastructure. Accessed February 3, 2021. <u>https://pubs.iied.org/10882IIED</u>.
- INDIS Indigenous Sustainable Development. Accessed February 3, 2021. https://indisproject.org/.
- International Development Committee, House of Commons, 2020. *The Newton Fund Review: Report of the Sub-Committee on the Work of ICAI: Government Response to the Committee's Third Report*. Accessed February 5, 2021. <u>https://publications.parliament.uk/pa/cm5801/cmselect/cmintdev/742/74202.htm</u>.

Itad & Rand, 2020. GCRF Evaluation, Inception Report. Accessed February 25, 2021.

Journal Periferias, n.d. GLOBAL GRACE. Accessed February 3, 2021.

https://revistaperiferias.org/en/materia/global-grace-2/.

- Kabeer, Naila, and Ramya Subrahmanian, 1997. *Institutions, relations and outcomes: Framework and tools for gender-aware planning*. Brighton: Institute of Development Studies. Accessed February 25, 2021.
- Kabeer, Naila, 1994. *Reversed Realities: Gender Hierarchies in Development Thought (1994)*. Accessed February 5, 2021. <u>https://nailakabeer.net/1994/12/27/reversed-realities-gender-hierarchies-in-</u> <u>development-thought-1994/</u>
- Korosteleva, Elena, and Zachary Paikin, 2020. *Russia between East and West, and the Future of Eurasian Order*. International Politics, July 14, 2020. Accessed February 25, 2021. <u>https://doi.org/10.1057/s41311-020-00261-5</u>.
- Krishnan, Aarti, Matt Alford, and Giovanni Pasquali, 2020. *Governing Value Chains in an Era of Polycentric Trade: Implications for Kenyan Horticultural First-Tier Suppliers*. SASE, 2020. Accessed February 25, 2021. <u>https://sase.confex.com/sase/2020/meetingapp.cgi/Paper/13375</u>.
- Lancaster University, 2019. ProtoPolicyAsia: empowering local communities and Government in Malaysia in addressing social issues in ageing and disabilities. Accessed February 3, 2021. http://imagination.lancaster.ac.uk/wp
 - content/uploads/2019/01/protopolicyasia_1_page_summary_1.pdf.
- Lee, Clarissa Ai Ling, and Sian Lun Lau, 2018. *Imagining Future Foods Through Speculative Design*. Accessed February 25, 2021.
- Loughborough University, 2019. Winners of SAFOD Self-Sufficiency design project design catalogue. Accessed February 3, 2021. <u>https://www.lboro.ac.uk/departments/design-school/whats-happening/news/2019/safod.html</u>.
- Matchett, Sara, and Phoebe Kisubi Mbasalaki, 2020. *Butoh Gives Back the Feeling to the People*. Agenda 34, no. 3: 74–86. Accessed February 25, 2021. <u>https://doi.org/10.1080/10130950.2020.1775102</u>.
- Mbasalaki, Phoebe Kisubi, 2020. Through the Lens of Modernity: Reflections on the (Colonial) Cultural Archive of Sexuality and Gender in South Africa. GLQ: A Journal of Lesbian and Gay Studies 26, no. 3: 455–75. Accessed February 25, 2021. <u>https://doi.org/10.1215/10642684-8311800</u>.
- McKinsey, 2008. Enduring Ideas: The 7-S Framework. Accessed February 5, 2021. <u>https://www.mckinsey.com/business-functions/strategy-and-corporate-finance/our-insights/enduring-ideas-the-7-s-framework</u>.
- MIDEQ Migration for Diversity and Equality, n.d. Accessed February 4, 2021. https://www.mideq.org.
- Nadvi, Khalid, Shane Godfrey, and Giovanni Pasquali, 2020. *The Rise and Architecture of Apparel Regional Value Chains in Sub-Saharan Africa*. SASE, 2020. Accessed February 25, 2021. https://sase.confex.com/sase/2020/meetingapp.cgi/Paper/13374.
- None in Three Research Centre for the Prevention of Gender-based Violence, n.d. Accessed February 3, 2021. http://www.noneinthree.org/.
- Oxfam Policy & Policy, 1999. A Guide to Gender-Analysis Frameworks. Accessed February 5, 2021. https://policy-practice.oxfam.org/resources/a-guide-to-gender-analysis-frameworks-115397/.
- Oxford Brookes University, 2020. Navigating from Education to Employment in Crisis and Uncertainty. Accessed February 3, 2021. <u>https://www.brookes.ac.uk/about-brookes/events/navigating-from-education-to-employment-in-crisis-and-uncertainty/</u>.
- Partnership for Conflict, Crime, & Security Research, n.d. *Projects Awarded under the ESRC-AHRC Forced Displacement Call*. Accessed February 3, 2021. <u>https://www.paccsresearch.org.uk/15-projects-under-the-esrc-ahrc-forced-displacement-call/</u>.
- Pasquali, Giovanni, and Shane Godfrey, 2020. *Covid-19 and the Governance of Regional Apparel Value Chains: Implications for Suppliers and Workers in Eswatini Global Development Institute*. Accessed February, 25. 2020.
- Pasquali, Giovanni, Shane Godfrey, and Khalid Nadvi, 2020. Understanding Regional Value Chains through the Interaction of Public and Private Governance: Insights from Southern Africa's Apparel Sector. Journal of International Business Policy. Accessed February 25, 2021. <u>https://doi.org/10.1057/s42214-020-00071-9</u>.
- Pasquali, Giovanni, and Aarti Krishnan, 2020. Do More Diversified Suppliers Rebound Faster than Concentrated Suppliers in Times of Shocks? Insights from Kenya. Accessed February 25, 2021. https://doi.org/10.35188/UNU-WIDER/WBN/2020-8.

People's Palace Projects, n.d. Accessed February 3, 2021. <u>https://peoplespalaceprojects.org.uk/en/</u>.

Petrikova, Ivica, 2019. *Food-Security Governance in India and Ethiopia: A Comparative Analysis*. Third World Quarterly 40, no. 4: 743–62. Accessed February 25, 2021.

https://doi.org/10.1080/01436597.2018.1538734.

- Pilgun, Elena, 2020. Good Governance and Effectiveness of Public Institutions in Post-Soviet Eurasia' GCRF COMPASS Early Career Researcher Training School Working Proceedings. Accessed February 25, 2021.
- Queen's Printer of Acts of Parliament, 2014. *International Development (Gender Equality) Act 2014*. Accessed February 5, 2021. <u>https://www.legislation.gov.uk/ukpga/2014/9/section/1/enacted</u>.
- Report for the University and College Union, n.d. *Impact of The Covid-19 pandemic on university finances*. Accessed February 5, 2021. <u>https://londoneconomics.co.uk/wp-content/uploads/2020/04/LE-Impact-of-Covid-19-on-university-finances-FINAL.pdf</u>.
- Royal Academy of Engineering, n.d. *Africa Prize Guidance Notes*. Accessed February 4, 2021. <u>https://www.raeng.org.uk/publications/other/africa-prize-guidance-notes-(1)</u>.
- Royal Academy of Engineering, n.d. *Competition Overview Agri-Tech Catalyst Round 8: Agriculture and Food Systems Innovation Innovation Funding Service*. Accessed February 4, 2021. <u>https://apply-for-innovation-funding.service.gov.uk/competition/334/overview</u>.
- Royal Geographical Society, n.d. *Field Research Programme*. Accessed February 3, 2021. <u>https://www.rgs.org/in-the-field/field-research-programme/</u>.
- Royal Academy of Engineering, 2017. *GCRF Africa Catalyst Tender*. Accessed February 3, 2021. <u>https://www.raeng.org.uk/getattachment/grants-and-prizes/international-research-and-collaborations/africa-catalyst/GCRF-Africa-Catalyst-Tender.pdf</u>

Royal Academy of Engineering, 2017. *Phase 2 Projects*. Accessed February 3, 2021. <u>https://www.raeng.org.uk/global/sustainable-development/africa-grants/africa-catalyst/current-recent-awards/phase-2-projects</u>.

- Royal Society, n.d. *Challenge-Led Grants*. Accessed February 4, 2021. <u>https://royalsociety.org/grants-schemes-awards/grants/challenge-led-grants/</u>.
- Royal Society, n.d. *FLAIR Fellowships*. Accessed February 4, 2021. <u>https://royalsociety.org/grants-schemes-awards/grants/flair/</u>.
- SAFOD, n.d. Accessed February 3, 2021. <u>https://www.safod.net/</u>.
- Salmanidou, Dimitra M., Mohammad Heidarzadeh, and Serge Guillas, 2019. *Probabilistic Landslide-Generated Tsunamis in the Indus Canyon, NW Indian Ocean, Using Statistical Emulation*. Pure and Applied Geophysics 176, no. 7: 3099–3114. <u>https://doi.org/10.1007/s00024-019-02187-3</u>.
- Science and Technology Facilities Council, 2018. *GCRF 2018 Applicant Guidance*. Accessed February 4, 2021. <u>https://stfc.ukri.org/files/gcrf-2018-applicant-guidance/</u>.
- Scotland's Rural College, n.d. *Formulating Value Chains for Orphan Crops in Africa*. Accessed February 4, 2021. <u>https://www.sruc.ac.uk/homepage/1123/formulating value chains for orphan crops in africa</u>.
- Statute Law Database, 2010. *Equality Act*. Accessed February 5, 2021. https://www.legislation.gov.uk/ukpga/2010/15/contents.

Tarana Khan Blog, n.d. Accessed February 4, 2021. <u>http://taranakhanauthor.com/publications/publications/</u>. Tetra Tech, 2020. *Review of Approaches to Gender Equality*. Accessed February 5, 2021.

- https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/90 8561/Review of Approaches to Gender Equality report.pdf.
- The Lancet, 2020. COVID-19 is a global health emergency that demands a global solution. No community is safe from SARS-CoV-2 unless all communities are protected. Accessed February 4, 2021. https://www.thelancet.com/journals/lancet/issue/vol396no10265/PIIS0140-6736(20)X0050-X.
- Tomorrow's Cities, n.d. Adapting Research Projects to the COVID-19 Pandemic: Experiences from the Nairobi Risk Hub. Accessed February 4, 2021. <u>https://tomorrowscities.org/adapting-research-projects-covid-19-pandemic-experiences-nairobi-risk-hub</u>.
- Tomorrow's Cities, n.d. Considering Volatile Livelihoods Is Critical in the War against the Covid-19 Disaster." Tomorrow's Cities. Accessed February 4, 2021. <u>https://tomorrowscities.org/considering-volatile-livelihoods-critical-war-against-covid-19-disaster-0</u>.
- Tomorrow's Cities, n.d. *Disasters, Humanitarianism and Emergencies: A Politics of Uncertainty*. Accessed February 4, 2021. <u>https://www.tomorrowscities.org/disasters-humanitarianism-and-emergencies-politics-uncertainty</u>.

- Tomorrow's Cities, 2019. *Mainstreaming Gender and Complexity: An Introduction for Research Design*. Accessed February 4, 2021. <u>https://tomorrowscities.org/mainstreaming-gender-and-complexity-introduction-research-design</u>.
- Tomorrow's Cities, 2020. Leaving No One behind in Tomorrow's Cities: Strengthening Gender, Intersectionality and Social Inclusion in the COVID-19 Crisis and beyond. Accessed February 4, 2021. <u>https://tomorrowscities.org/leaving-no-one-behind-tomorrows-cities-strengthening-gender-intersectionality-and-social-inclusion</u>.
- Tomorrow's Cities, 2020. Strengthening Community-Based Disaster Management Institutions to Tackle COVID-19 and Local Disasters. Accessed February 4, 2021. <u>https://www.tomorrowscities.org/strengthening-community-based-disaster-management-institutions-tackle-covid-19-and-local-disasters.</u>
- Tomorrow's Cities, 2020. *The COVID-19 Pandemic and Its Challenges for Ecuador*. Accessed February 4, 2021. https://www.tomorrowscities.org/covid-19-pandemic-and-its-challenges-ecuador.
- Tomorrow's Cities, 2020. *Tomorrow's Cities and Covid-19: A Discussion*. Accessed February 4, 2021. <u>https://tomorrowscities.org/tomorrows-cities-and-covid-19-discussion-0</u>.
- Tomorrow's Cities, 2020. *Tomorrow's Cities Safeguarding Policy*. Accessed February 4, 2021. https://www.tomorrowscities.org/tomorrows-cities-safeguarding-policy.
- Tomorrow's Cities, 2020. Trend of Urban Growth in Nepal with a Focus in Kathmandu Valley: A Review of Processes and Drivers of Change. Accessed February 4, 2021. <u>https://www.tomorrowscities.org/trend-urban-growth-nepal-focus-kathmandu-valley-review-processes-and-drivers-change</u>.
- Tsekleves, Emmanuel, Andy Darby, Clarissa Ai Ling Lee, Siti Ismail, and Min Hooi Yong, 2020. *The Little Book of Speculative Design for Policy-Makers in Malaysia*. Accessed February 25, 2021.
- Tsekleves, Emmanuel, Andy Darby, Clarissa Ai Ling Lee, and Min Hooi Yong, 2020. *The Little Book of Speculative Design for NGOs and Community Groups in Malaysia*. Accessed February 25. 2021.
- UCL Institute for Risk and Disaster Reduction, 2018. MANTRA: Increasing Maternal and Child Health Resilience before, during and after Disasters Using Mobile Technology in Nepal. Accessed February 24, 2021. <u>https://www.ucl.ac.uk/risk-disaster-reduction/mantra-increasing-maternal-and-child-health-resilienceduring-and-after-disasters-using-mobile</u>.
- UIPE The Uganda Institution of Professional Engineers, n.d. Accessed February 3, 2021. <u>http://www.uipe.co.ug/</u>.
- UIPE, 2020. UIPE Newseltter July-September2020. Accessed February 3, 2021. http://www.uipe.co.ug/files/downloads/UIPE%20Newseltter%20July-September2020.pdf.
- United Nations, n.d. *Chapter I: Identifying social inclusion and exclusion*. Accessed February 5, 2021. <u>https://www.un.org/esa/socdev/rwss/2016/chapter1.pdf</u>.
- United Nations, 2018. Ending Poverty. Accessed February 24, 2021. <u>https://www.un.org/en/sections/issues-depth/poverty/</u>.
- UK Collaborative on Development Research, 2019. GCRF Global Multimorbidity Seed-Funding 2019. Accessed February 4, 2021. <u>https://www.ukcdr.org.uk/funding-call/gcrf-global-multimorbidity-seed-funding-2019/</u>.
- UK Parliament, 2020. *Spending Review: Reducing the 0.7% Aid Commitment*. Accessed February 24, 2021. https://commonslibrary.parliament.uk/spending-review-reducing-the-aid-commitment/.
- UKRI, n.d. Diversity Data. Accessed February 5, 2021. <u>https://www.ukri.org/our-work/supporting-healthy-research-and-innovation-culture/equality-diversity-and-inclusion/diversity-data/</u>.
- University of Huddersfield Research Portal, n.d. *None In Three A Centre for the Development, Application, Research and Evaluation of Prosocial Games for the Prevention of Gender-Based Violence*. Accessed February 3, 2021. <u>https://pure.hud.ac.uk/en/projects/none-in-three-a-centre-for-the-development-application-research-a</u>.
- University of Kent, n.d. *Newsletters GCRF Compass*. Accessed February 3, 2021. <u>https://research.kent.ac.uk/gcrf-compass/newsletters/</u>.
- Vashisht, Devika, Amit Kumar, Surinder Kumar Mehta, and Alex Ibhadon, 2020. Analysis of Emerging Contaminants: A Case Study of the Underground and Drinking Water Samples in Chandigarh, India. Environmental Advances 1: 100002. Accessed February 25, 2020. <u>https://doi.org/10.1016/j.envadv.2020.100002</u>.
- Zaman, Shamrita, Peter Sammonds, Bayes Ahmed, and Taifur Rahman, 2020. Disaster Risk Reduction in Conflict Contexts: Lessons Learned from the Lived Experiences of Rohingya Refugees in Cox's Bazar,

Bangladesh. International Journal of Disaster Risk Reduction 50: 101694. Accessed February 25, 2020. <u>https://doi.org/10.1016/j.ijdrr.2020.101694</u>.

DP and Award Level/Internal

Academy of Medical Sciences, n.d. AMS Networking Grants. Accessed February 5, 2021. Academy of Medical Sciences, 2018. Annual Diversity Report 2018-2019. Accessed February 5, 2021. Academy of Medical Sciences, 2018. Annual Report 2018-2019. Accessed February 5, 2021. Academy of Medical Science, n.d. GCRF Round 7 Scheme Notes. Accessed February 5, 2021. Academy of Medical Sciences, n.d Mental Health Case Study. Accessed February 5, 2021. Academy of Medical Sciences, n.d. Multimorbidity Case Study. Accessed February 5, 2021. Academy of Medical Sciences, n.d. Networking Grant Panel Guidance Round 6. Accessed February 5, 2021. Academy of Medical Sciences, n.d. Rapid Diagnostic Case Study. Accessed February 5, 2021. Academy of Medical Sciences, 2020. Understanding the Food Chain of Teff: AMS Letter about Coronavirus. Accessed February 4, 2021. Academy of Medical Sciences, 2020. Understanding the Food Chain of Teff: Covid Extension Request. Accessed February 4, 2021. Academy of Medical Sciences, 2020. Understanding the Food Chain of Teff: No Cost Extension Agreement. Accessed February 4, 2021. Academy of Medical Sciences, 2020. Understanding the Food Chain of Teff: MoU DBU RHUL. Accessed February 4, 2021. Academy of Medical Sciences, 2020. Understanding the Food Chain of Teff: Programme GCRF Workshop. Accessed February 4, 2021. AHRC, n.d. AHRC and the GCRF Areas Circle Strategy. Accessed February 5, 2021. AHRC, 2019. AHRC Delivery Plan 2019. Accessed February 5, 2021. AHRC, 2017. AHRC ESRC GCRF Forced Displacement Project. Access February 5, 2021. AHRC, n.d. AHRC GCRF Area Focused Network Plus EOI Call Document. Accessed February 5, 2021. AHRC, 2020. AHRC GCRF FoF Highlight 2020. Accessed February 5, 2021 AHRC, n.d. Disability and Embodiment in Namibia: CfS. Accessed February 4, 2021. AHRC, n.d. Disability and Embodiment in Namibia: Gender Equality Statement. Accessed February 4, 2021. AHRC, n.d. Disability and Embodiment in Namibia: ODA Compliance Statement. Accessed February 4, 2021. AHRC, n.d. Disability and Embodiment in Namibia: Pathways. Accessed February 4, 2021. AHRC, n.d. GCRF Coherence Food Culture Call. Accessed February 5, 2021. AHRC, n.d. Gender Justice Hub. Accessed February 5, 2021. AHRC, 2018. GlobalGRACE Assurance Documents. Accessed February 3, 2021. AHRC, n.d. GlobalGRACE Collaborative Agreement. Accessed February 3, 2021. AHRC, 2017. GlobalGRACE Contextual and Problem Analysis. Accessed February 3, 2021. AHRC, 2017. GlobalGRACE Detailed Pathways to Impact. Accessed February 3, 2021. AHRC, 2017. GlobalGRACE Proposal and Application. Accessed February 3, 2021. AHRC, 2019. GlobalGRACE Report by Independent Evaluator Srila Roy. Accessed February 3, 2021. AHRC, 2019. GlobalGRACE Report by Independent Evaluator Susan Stanford Friedman. Accessed February 3, 2021. AHRC, 2018. GlobalGRACE Stage Gate Review - Feedback from GCRF. Accessed February 3, 2021. AHRC, n.d. Raising awareness of inclusive crafts: building capacity and self-sufficiency of communication within the Charity SAFOD and the Countries of SA: CfS. Accessed February 4, 2021. AHRC, n.d. Raising awareness of inclusive crafts: building capacity and self-sufficiency of communication within the Charity SAFOD and the Countries of SA: DevTracker Project. Accessed February 3, 2021. https://devtracker.fcdo.gov.uk/projects/GB-GOV-13-FUND--GCRF-AH S005765 1. AHRC, n.d. Raising awareness of inclusive crafts: building capacity and self-sufficiency of communication within the Charity SAFOD and the Countries of SA: ODA Statement. Accessed February 4, 2021. AHRC, n.d. UKRI AHRC GCRF Urgency Grants Scheme Highlight Notice for Innovation. Accessed February 5, 2021. AHRC, n.d. UKRI GCRF Collective Programme – AHRC Network Plus Calls. Accessed February 24, 2021. BBSCR, 2016, n.d. BBSCR Impact Report. Accessed February 5, 2021.

BBSCR, 2019. BBSRC Delivery Plan 2019. Accessed February 5, 2021.

BBSCR, 2017. BBSCR Impact Report. Accessed February 5, 2021.

- BBSCR, 2018. Child Health, Agriculture and Integrated Nutrition (CHAIN): A Randomized Trial to Close the Nutrient Gap in Rural Zimbabwe Case for Support. Accessed February 4, 2021.
- BBSCR, n.d. Equality Impact Assessment SPF Food System CDT. Accessed February 5, 2021.
- BBSCR, n.d. Equality Impact Assessment BPD. Accessed February 5, 2021.
- BBSCR, 2018. Food & nutrition research for health in the developing world: bioavailability and nutrient content ODA Compliance. Accessed February 4, 2021.
- BBSCR, 2018. Food & nutrition research for health in the developing world: bioavailability and nutrient content Pathways to Impact. Accessed February 4, 2021.

BBSCR, n.d. Collective Programme Double Burden Malnutrition Call Guidance. Accessed February 5, 2021.

- BBSCR, n.d. One Health Poultry Hub. Accessed February 5, 2021.
- BBSCR, n.d. The Phenomenal Growth of the Poultry Layer Industry. Accessed February 5, 2021.
- British Academy, n.d. CI Assessment Guidelines. Accessed February 5, 2021.
- British Academy, n.d. Cl170378 Final Report. Accessed February 5, 2021.
- British Academy, n.d. Cl170378 First Interim Report. Accessed February 5, 2021.
- British Academy, n.d. Cl170378 Second Interim Report. Accessed February 5, 2021.
- British Academy, n.d. Cl170378 Third Interim Report. Accessed February 5, 2021.
- British Academy, n.d. Cities and Infrastructure Scheme Notes. Accessed February 5, 2021.
- British Academy, n.d. *Code of Practice*. Accessed February 5, 2021.
- British Academy, n.d. Early Childhood Development Scheme Notes. Accessed February 5, 2021.
- British Academy, n.d. Early Childhood Development Successful Applications. Accessed February 5, 2021.
- British Academy, n.d. *Early Childhood Development Terms & Conditions*. Accessed February 5, 2021.
- British Academy, n.d. EC170232 Final Report. Accessed February 5, 2021.
- British Academy, n.d. EC170232 Interim Report 1. Accessed February 5, 2021.
- British Academy, n.d. EC170232 Interim Report 2. Accessed February 5, 2021.
- British Academy, n.d. EC170232 Interim Report 3. Accessed February 5, 2021.
- British Academy, 2019. ECE 2019 Peer Review Guidelines. Accessed February 5, 2021.
- British Academy, 2019. ELC 2019 Peer Review Guidelines. Accessed February 5, 2021.
- British Academy, n.d. ELC Scheme Notes. Accessed February 5, 2021.
- British Academy, n.d. ELC Successful Applications. Accessed February 5, 2021.
- British Academy, n.d. ELC T&Cs. Accessed February 5, 2021.
- British Academy, n.d. Feedback from Learning workshop. Accessed February 5, 2021.
- British Academy, n.d. GCRF Early Child Call Specifications. Accessed February 5, 2021.
- British Academy, 2017. GCRF Networking Grants. Accessed February 5, 2021.
- British Academy, n.d. HDV190030 Full Proposal. Accessed February 4, 2021.
- British Academy, n.d. HDV190030 Interim Report 1. Accessed February 5, 2021.
- British Academy, n.d. *HDV190030 Collaboration Agreement*. Accessed February 4, 2021.
- British Academy, n.d. HDV190030 Ethics Revised Version Final. Accessed February 4, 2021.
- British Academy, n.d. HDV190030 Health and Safety Risk Assessment. Accessed February 4, 2021.
- British Academy, n.d. *HDV190030 Governance, Field Work, and Preliminary Findings Report*. Accessed February 4, 2021.
- British Academy, n.d. HDV190030 No Cost Extension Form. Accessed February 4, 2021.
- British Academy, n.d. HDV190030 Literature Review 1 Resistance to VAWG. Accessed February 4, 2021.
- British Academy, n.d. HDV190030 Literature Review 2 Activism and Women's Police Stations. Accessed
 - February 4, 2021.
- British Academy, 2019. HDV 2019 Peer Review Guidelines. Accessed February 5, 2021.
- British Academy, n.d. HDV Scheme Notes. Accessed February 5, 2021.
- British Academy, n.d. HDV Successful Applications. Accessed February 5, 2021.
- British Academy, n.d. HDV T&Cs. Accessed February 5, 2021.
- British Academy, 2018. Strategic Plan 2018 2022. Accessed February 5, 2021.
- British Academy, n.d. SDP2 100029 Interim Report 1. Accessed February 5, 2021.
- British Academy, n.d. SDP2 100222 Interim Report 1. Accessed February 5, 2021.
- British Academy, 2018. SDP 2018 Peer Review Guidelines. Accessed February 5, 2021.
- British Academy, n.d. SDP Scheme Notes. Accessed February 5, 2021.

British Academy, n.d. SDP2 Scheme Notes. Accessed February 5, 2021.

British Academy, n.d. SDP Successful Applications. Accessed February 5, 2021.

British Academy, n.d. SDP2 Successful Applications. Accessed February 5, 2021.

British Academy, 2016. *Sustainable Development Programme Scheme Notes 2016*. Accessed February 5, 2021. British Academy, n.d. *SDP T&Cs*. Accessed February 5, 2021.

British Academy, n.d. The gendered price of precarity: Workplace sexual harassment and young women's agency: Youth Futures Proposal. Accessed February 4, 2021. British Academy, 2020. The gendered price of precarity: Workplace sexual harassment and young women's agency: Narrative Report No 1 - Sept 2020. Accessed February 4, 2021. British Academy, n.d. Urban Infrastructures of Well-Being Scheme Notes. Accessed February 5, 2021. British Academy, n.d. Youth Futures Scheme Notes. Accessed February 5, 2021. Department for the Economy, 2018. GCRF Final Report. Accessed February 5, 2021. EPSRC, n.d. APSISSFE: Case Support. Accessed February 4, 2021. EPSRC, n.d. APSISSFE: Gender Equality Statement. Accessed February 4, 2021. EPSRC, n.d. APSISSFE: ODA Statement. Accessed February 4, 2021. EPSRC, n.d. APSISSFE: Pathways to Impact. Accessed February 4, 2021. ESRC, n.d. Citizenship Futures: Completed Proposal Form. Accessed February 4, 2021. ESRC, n.d. Citizenship Futures: ODA Compliance. Accessed February 4, 2021. ESRC, n.d. Citizenship Futures: Pathways to Impact. Accessed February 4, 2021. ESRC, n.d. Citizenship Futures: CFS. Accessed February 4, 2021. ESRC, 2019. Decent Work in Regional Value Chains: Case for Support. Accessed February 4, 2021. ESRC, 2019. Decent Work in Regional Value Chains: Concept Note - Garment. Accessed February 4, 2021. ESRC, 2019. Decent Work in Regional Value Chains: Concept Note - Horticulture. Accessed February 4, 2021. ESRC, 2019. Decent Work in Regional Value Chains: Concept Note - National Legilsation. Accessed February 4, 2021. ESRC, 2019. Decent Work in Regional Value Chains: Concept Note - Trade Flows. Accessed February 4, 2021. ESRC, 2019. Decent Work in Regional Value Chains: ODA statement. Accessed February 4, 2021. ESRC, 2019. Decent Work in Regional Value Chains: Pathways to Impact. Accessed February 4, 2021. ESRC, n.d. Development Based Approaches to Displacement. Accessed February 5, 2021. ESRC, 2017. Drugs and (dis)order: Case for Support. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Completed Proposal Form. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - Alcis. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - AREU. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - Christian Aid. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - KRC. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - LSE. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - LSHTM. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - National Uni of Colombia. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - OSDR. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - PosNeg. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - SHAN. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - SOAS. Accessed February 4, 2021. ESRC, 2017. Drugs and (dis)order: Letter of Support - University of Lost Andes. Accessed February 4, 2021. ESRC, n.d. Education as a Driver of Sustainable Development Call for Research Grants Call Specification. Accessed February 24, 2021. ESRC, n.d. Education as a Driver of Sustainable Development. Accessed February 5, 2021. ESRC, 2019. ESRC Delivery Plan 2019. Accessed February 5, 2021. ESRC, n.d. ESRC UKRI GCRF (Re)Thinking the Off-Grid City Call Specification. Accessed February 5, 2021. ESRC, n.d. From education to employment? Trajectories for young people in Lebanon in the context of

protracted displacement: Case for Support. Accessed February 4, 2021. ESRC, n.d. From education to employment? Trajectories for young people in Lebanon in the context of protracted displacement: ODA Compliance Statement. Accessed February 4, 2021.

ESRC, n.d. From education to employment? Trajectories for young people in Lebanon in the context of protracted displacement: Pathways to Impact. Accessed February 4, 2021.

ESRC, n.d. GCRF Climate Resilience KM Call Specification. Accessed February 5, 2021.

- ESRC, n.d. GCRF Inclusive Societies: Case for Support. Accessed February 4, 2021.
- ESRC, n.d. GCRF Inclusive Societies: ODA Compliance Statement. Accessed February 4, 2021.
- ESRC, n.d. GCRF Inclusive Societies: Pathways to Impact. Accessed February 4, 2021.
- ESRC, n.d. GCRF Network Plus Gender Call Specification. Accessed February 5, 2021.
- ESRC, 2019. GRRIPP: Case for Support. Accessed February 4, 2021.
- ESRC, n.d. GRRIPP: ODA Compliance Statement. Accessed February 4, 2021.
- ESRC, n.d. GRRIPP: Pathways to Impact. Accessed February 4, 2021.
- ESRC, n.d. *ReGHID ODA Compliance Statement*. Accessed February 4, 2021.
- ESRC, n.d. ReGHID Pathways to Impact. Accessed February 4, 2021.
- ESRC, n.d. UKRI GCRF Collective Programme Climate Resilience Network Information. Accessed February 5, 2021.
- ESRC, n.d. UKRI GCRF Education and Gender Network Plus. Accessed February 5, 2021.
- ESRC, n.d. UKRI GCRF Equitable Resilience Call Specification. Accessed February 5, 2021.
- ESRC, n.d. UKRI-GCRF Sustainable Energy Call Specification. Accessed February 5, 2021.
- HEFCW, n.d. Annual Report to BEIS. Accessed February 5, 2021.
- HEFCW, n.d. Final Strategy Aberystwyth. Accessed February 5, 2021.
- HEFCW, n.d. Final Strategy Bangor. Accessed February 5, 2021.
- HEFCW, n.d. Final Strategy Cardiff University. Accessed February 5, 2021.
- HEFCW, n.d. Final Strategy Swansea University. Accessed February 5, 2021.
- HEFCW, 2020. GCRF Funding MoU 2020-21. Accessed February 5, 2021.
- HEFCW, 2020. Research and Innovation the Vision for Wales. Accessed February 5, 2021.
- HEFCW, 2020. W20 16HE Global Challenges Research Fund 2020/21. Accessed February 5, 2021.
- Innovate UK, n.d. Demonstrate Impact Profiles Round 1 Phase 1. Accessed February 5, 2021.
- Innovate UK, n.d. *Demonstrate Impact: using HCD to Innovate*. Accessed February 5, 2021.
- Innovate UK, n.d. GCRF I&C Bid InnovateUK Demonstrate for Impact. Accessed February 5, 2021.
- Innovate UK, n.d. GCRF I&C Bid Innovate UK Energy. Accessed February 5, 2021.
- Innovate UK, 2019. Global Young Innovators Programme. Accessed February 5, 2021.
- Innovate UK, n.d. Innovation Beyond Energy Catalyst. Accessed February 5, 2021.
- Innovate UK, 2020. NutriNuts Monitoring Report. Accessed February 4, 2021.
- Innovate UK, n.d. Nutri Nuts Project Presentation. Accessed February 4, 2021.
- Innovate UK, n.d. ULA Theory of Change Final. Accessed February 5, 2021.
- Innovate UK, n.d. Urban Links Africa Programme. Accessed February 5, 2021.
- MRC, 2018. Annual Report and Accounts 2018-19. Accessed February 24, 2021.
- MRC, n.d. *Call for research in Global Multimorbidity Seed-funding Grant Application Guidance*. Accessed February 5, 2021.
- MRC, 2018. MRC AR 2018-19. Accessed February 5, 2021.
- MRC, n.d. Building resilience and resources to overcome depression and anxiety in young people from urban neighbourhoods in Latin America: OLA Grant Application. Accessed February 3, 2021.
- MRC, 2019. MRC Delivery Plan 2019. Accessed February 5, 2021.
- MRC, 2014. MRC Strategic Plan 2014-2019. Accessed February 5, 2021.
- MRC, n.d. Parental interventions to support families of children with neurodisabilities in low-resource settings: Newton MRC Wellcome Trust Development Proposal. Accessed February 4, 2021.
- MRC, n.d. Parental interventions to support families of children with neurodisabilities in low-resource settings: Pathways to Impact Newton. Accessed February 4, 2021.
- MRC, n.d. The PRECISE Network: Project Parnter Letter of Support LGT. Accessed February 4, 2021.
- MRC, n.d. The PRECISE Network: Project Parnter Letter of Support Mircolife. Accessed February 4, 2021.
- MRC, n.d. The PRECISE Network: Project Parnter Letter of Support QuantuMD. Accessed February 4, 2021.
- MRC, n.d. The PRECISE Network: Project Parnter Letter of Support -Quanys. Accessed February 4, 2021.
- MRC, n.d. *Menstrual cups and unconditional cash transfer to reduce sexual and reproductive harm and school drop-out in adolescent schoolgirls in western Kenya (Cups or Cash)* MOU Confidentiality Agreement Signed. Accessed February 25, 2021.
- MRC, n.d. Cups or Cash MOE County Support. Accessed February 25, 2021.
- MRC, n.d. Cups or Cash Pathways to Impact. Accessed February 25, 2021.

MRC, 2020. Cups or Cash Protocol. Accessed February 25, 2021.

MRC, n.d. Cups or Cash MOE Endorsement Letter JGHT CoC. Accessed February 25, 2021.

MRC, n.d. Cups or Cash SWAP Final Contract Signed. Accessed February 25, 2021.

NERC, 2018. ESPA 2018 Interdisciplinary Research for Development Impact. Accessed February 5, 2021.

NERC, 2018. ESPA 2018 The Role of Equitable Partnerships. Accessed February 5, 2021.

NERC, n.d. MANTRA: Case for Support. Accessed February 4, 2021.

NERC, n.d. MANTRA: Completed Proposal Form. Accessed February 4, 2021.

NERC, n.d. MANTRA: Letter of Support - MFD. Accessed February 4, 2021.

NERC, n.d. MANTRA: Letter of Support – PhotoCircle. Accessed February 4, 2021.

NERC, n.d. MANTRA: Pathways to Impact. Accessed February 4, 2021.

NERC, 2019. NERC Delivery Plan 2019. Accessed February 5, 2021.

NERC, n.d. Tomorrow's Cities and Covid-19 A Discussion. Accessed February 5, 2021.

NERC, n.d. Tomorrow's Cities: GCRF Urban Disaster Risk Hub Case for Support. Accessed February 4, 2021.

NERC, n.d. Tomorrow's Cities: GCRF Urban Disaster Risk Hub Completed Proposal Form. Accessed February 4, 2021.

NERC, n.d. Tomorrow's Cities: GCRF Urban Disaster Risk Hub Pathways to Impact. Accessed February 4, 2021.

NERC, n.d. UKRI GCRF Collective Programme - Multihazards Call. Accessed February 5, 2021.

NERC, n.d. Urban Disaster Hub Program. Accessed February 5, 2021.

Research England, n.d. Aberdeen GCRF Strategy. Accessed February 5, 2021.

Research England, n.d. Abertay GCRF Strategy. Accessed February 5, 2021.

Research England, 2017. AY 2017-18 QR GCRF Monitoring Guidelines. Accessed February 5, 2021.

Research England, 2018. AY 2018-19 QR GCRF Monitoring Guidance. Accessed February 5, 2021.

Research England, n.d. Institute of Cancer Research QR GCRF Strategy. Accessed February 5, 2021.

Research England, 2019. Research England Delivery Plan 2019. Accessed February 5, 2021.

Research England, 2018. Research England Monitoring Report to BEIS - 2018-19. Accessed February 5, 2021.

Research England, n.d. Royal College of Music QR GCRF Strategy. Accessed February 5, 2021.

Research England, n.d. Royal Holloway University QR GCRF Strategy. Accessed February 5, 2021.

Research England, n.d. Terms and Conditions of Research England QR GCRF Grant. Accessed February 5, 2021.

Research England, n.d. Queens University GCRF Strategy. Accessed February 5, 2021.

Research England, 2017. QR Strategic Policy. Accessed February 5, 2021.

Research England, n.d. Ulster University GCRF Strategy. Accessed February 5, 2021.

Research England, n.d. University of Bristol QR GCRF Strategy. Accessed February 5, 2021.

Research England, n.d. University of Exeter QR GCRF Strategy. Accessed February 5, 2021.

Research England, n.d. University of Hull Annex E - Case Study - DEVELOP. Accessed February 5, 2021.

Research England, n.d. University of Lincoln QR GCRF Strategy. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. Academy Statement and Action Plan Researcher Development Concordat. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. Africa Catalyst Inception Report. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. *Africa Innovation Fellowship*. Accessed February 24, 2021.

Royal Academy of Engineering, 2021. Africa Prize for Engineering Innovation. Accessed February 5, 2021.

Royal Academy of Engineering, 2019. Africa Prize Final Report. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. Africa Prize Final Report (CiTi - Y4). Accessed February 5, 2021.

Royal Academy of Engineering, n.d. Agri Tech Catalyst Log Frame. Accessed February 5, 2021.

Royal Academy of Engineering, 2019. AIF Fellows Insight Report. Accessed February 5, 2021.

Royal Academy of Engineering, 2020. AIF Innovators Insight Report. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. Alumni Grants Guidance for Reviewers. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. Alumni Grants Guidance Notes. Accessed February 5, 2021.

Royal Academy of Engineering, 2019. Annual Report. Accessed February 5, 2021.

Royal Academy of Engineering, 2019. AP Stats. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. AP Year 7 Application Guidance Notes. Accessed February 5, 2021.

Royal Academy of Engineering, n.d. APAlumni20 EiE. Accessed February 5, 2021.

Royal Academy of Engineering, 2020. *Applicant Guidance for Research Fellowships 2020-21*. Accessed February 5, 2021.

Royal Academy of Engineering, 2017. Application Form (J Shen). Accessed February 5, 2021.

Royal Academy of Engineering, 2020. BIA Post Event Insight. Accessed February 5, 2021.

- Royal Academy of Engineering, n.d. Bullying and Harassment Policy. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. DFID ATC Additional Monitoring. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. EaBW Core Guide. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. *Eco-Engineered Biofilters for Sustainable Removal of Pesticides in Drinking Water*. Accessed February 25, 2021.
- Royal Academy of Engineering, n.d. Ecorys Annual Review. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. *Engineering X Pandemic Preparedness Grant Scheme Applicant Guidance*. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. *Engineering X Pandemic Reponse Reviewer Guidance*. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Equality and Diversity Policy. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. EXPPP Governance. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Feedback Report FoD1 Kigali. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. FINAL Bid with Case Study Agri-Tech Catalyst. Accessed February 5, 2021.
- Royal Academy of Engineering, 2016. Final Financial Report. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d Final Report GGCS Mexico. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Follow-on Funding Guidance for Applicants. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Follow-on Guidance for Reviewers. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. GCRF AgriFood Africa Innovation Awards Case for Support Template. Accessed February 5, 2021
- Royal Academy of Engineering, n.d. *GCRF Africa Catalyst Sustainable Infrastructure Applicant Guidance Notes*. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. *GCRF Africa Catalyst Sustainable Infrastructure Reviewer Guidance Notes*. Accessed February 5, 2021.
- Royal Academy of Engineering, 2019. GCRF Annual Report 19-20 AgriFood Africa. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. *GCRF ATC Update*. Accessed February 5, 2021.
- Royal Academy of Engineering, 2019. GGCS 2019 Brochure. Accessed February 5, 2021.
- Royal Academy of Engineering, 2019. HEP SSA 2019-2021 Applicant Guidance Notes 2.1. Accessed February 5, 2021.
- Royal Academy of Engineering, 2019. HEP SSA Reviewer Guidance Notes 2019. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. HEPSSA1-4. Accessed February 5, 2021.
- Royal Academy of Engineering, 2019. KTN Global Alliance Annual Report FY 19-20. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Matovu About the Vertical Farm. Accessed February 5, 2021.
- Royal Academy of Engineering, 2019. Mid-Year Reporting Form. Accessed February 5, 2021.
- Royal Academy of Engineering, 2015. ODA Strategic and Finance Business Case. Accessed February 5, 2021.
- Royal Academy of Engineering, 2019. Online Mapping Report. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Project CARE Theme 3 Guidance for Reviewers. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Project CARE Theme 3 Letter to Entrepreneurs. Accessed February 5, 2021.
- Royal Academy of Engineering, 2020. RAE FINAL REPORT 2020. Accessed February 5, 2021.
- Royal Academy of Engineering, 2018. RAEng Anti Bribery Policy. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. RAEng GCRF Additional Bid 020916. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. RAEng GCRF Draft Delivery Plan. Accessed February 5, 2021.
- Royal Academy of Engineering, 2018. RAEng Modern Slavery Policy Statement. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. RAEng Privacy Policy General Notice. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Reviewer Guidance Notes Tranche 4. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Reviewer Guidance Notes Tranche 9. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Safeguarding Policy. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Satellite Event Guidance Notes. Accessed February 5, 2021.
- Royal Academy of Engineering, 2018. Terms of Reference HEP SSA. Accessed February 5, 2021.
- Royal Academy of Engineering, n.d. Theory of Change Summary. Accessed February 5, 2021.
- Royal Society, 2018. CHL Research Proposal. Accessed February 5, 2021.
- Royal Society, n.d. FLAIR TOC. Accessed February 5, 2021.
- Royal Society, n.d. RS Aggregated Diversity Data Ethnicity and Disability. Accessed February 5, 2021.
- Royal Society, n.d. RS Aggregated Diversity Data Gender. Accessed February 5, 2021.

Royal Society, 2020. Digital soil information for targeting soil, water and nutrient management options for enhanced crop yields and livelihoods in Uganda: FLAIR Fellowships Award. Accessed February 4, 2021.

Scottish Funding Council, n.d. *Annex A1 Template for Institutional Three-Year GC*. Accessed February 5, 2021. Scottish Funding Council, n.d. *Dundee GCRF Strategy*. Accessed February 5, 2021.

Scottish Funding Council, n.d. Edinburgh Napier University GCRF Strategy. Accessed February 5, 2021.

Scottish Funding Council, n.d. Edinburgh University GCRF Strategy. Accessed February 5, 2021.

Scottish Funding Council, n.d. *Formulating Value Chains for Orphan Crops in Africa: Standard Proposal.* Accessed February 4, 2021.

Scottish Funding Council, n.d. *Formulating Value Chains for Orphan Crops in Africa: Collaborator Details*. Accessed February 25, 2021.

Scottish Funding Council, n.d. Glasgow School of Art GCRF Strategy. Accessed February 5, 2021.

Scottish Funding Council, n.d. *Glasgow University GCRF Strategy*. Accessed February 5, 2021.

Scottish Funding Council, 2019. GCRF Case Studies. Accessed February 5, 2021.

Scottish Funding Council, n.d. GCRF Gender Equality Statement Toolkit. Accessed February 5, 2021.

Scotting Funding Council, n.d. RCS GCRF Strategy. Accessed February 5, 2021.

Scotting Funding council, n.d. RGU GCRF Strategy. Accessed February 5, 2021.

Scotting Funding Council, n.d. SRUC GCRF Strategy. Accessed February 5, 2021.

Scotting Funding Council, n.d. Stirling GCRF Strategy. Pdf. Accessed February 5, 2021.

Scotting Funding Council, n.d. Strathclyde GCRF Strategy. Accessed February 5, 2021.

Scottish Funding Council, n.d. UWS GCRF Strategy. Accessed February 5, 2021.

Scottish Funding Council, 2019. GCRF October 2019 SFC Reporting. Accessed February 5, 2021.

Scottish Funding Council, n.d. GCU GCRF Strategy. Accessed February 5, 2021.

Science and Technology Facilities Council, n.d. TRANSSITioN Case of Support. Accessed February 4, 2021.

Science and Technology Facilities Council, n.d. TRANSSITioN Proposal. Accessed February 4, 2021.

Science and Technology Facilities Council, n.d. TRANSSITioN ODA STFC GCRF with Summary. Accessed February 4, 2021.

Science and Technology Facilities Council, n.d. *TRANSSITioN Pathway to Impact*. Accessed February 4, 2021. UKRI, n.d. *Accelerate Hub Findings*. Accessed February 5, 2021.

UKRI, n.d. Digital Innovation for Development in Africa (DIDA) Call. Accessed February 5, 2021.

UKRI, n.d. GCRF Demonstrate Impact Theory of Change. Accessed February 5, 2021.

UKRI, n.d. GCRF GROW GC Call. Accessed February 5, 2021.

UKRI, n.d. GCRF Hubs Call. Accessed February 5, 2021.

UKRI, n.d. GCRF Networking Grant Award Data. Accessed February 5, 2021.

UKRI, n.d. GCRF ODA Guidance. Accessed February 5, 2021.

UKRI, n.d. GCRF ODA Guidance for Challenge Clusters. Accessed February 5, 2021.

UKRI, n.d. GCRF Reporting Template Year 2 of 3. Accessed February 5, 2021.

UKRI, 2019. UKRI Delivery Plan 2019. Accessed February 5, 2021.

UKRI, n.d. UKRI International Development Peer Review College. Accessed February 5, 2021.

UK Space Agency, n.d. Agriompas Proposal. Accessed February 5, 2021.

UK Space Agency, n.d. Assessment Sheet for IPP Call 3 INDEPENDENT ASSESSORS. Accessed February 5, 2021.

UK Space Agency, n.d. Assessment Sheet for IPP Call 3 UKSA ASSESSORS. Accessed February 5, 2021.

UK Space Agency, n.d. Assessment Sheet for Caribou IPP Call 3. Accessed February 5, 2021.

UK Space Agency, n.d. *IPP Assessors TOR Conflict of Interest and Confidentiality Agreement - Call 3*. Accessed February 5, 2021.

UK Space Agency, n.d. IPP Call 3 Application Form. Accessed February 5, 2021.

UK Space Agency, n.d. IPP Call 2 Application Guidance. Accessed February 5, 2021.

UK Space Agency, n.d. IPP Call 3 Application Guidance. Accessed February 5, 2021.

UK Space Agency, n.d. IPP Gender Strategy Final. Accessed February 5, 2021.

UK Space Agency, 2020. IPP Logframe. Accessed February 5, 2021.

UK Space Agency, n.d. IPP M&E Handbook. Accessed February 5, 2021.

UK Space Agency, 2020. IPP M&E Plan Revised. Accessed February 5, 2021.

UK Space Agency, 2020. IPP Projects for GCRF. Accessed February 5, 2021.

UK Space Agency, n.d. UKSA SPACE UK Solutions for Agriculture. Accessed February 5, 2021.

UK Space Agency, n.d. UKSA Disaster Resilience. Accessed February 5, 2021.

UK Space Agency, n.d. UKSA UK Space Sector Catalogue. Accessed February 5, 2021.

UK Space Agency, 2019. UKSA Finance Report. Accessed February 5, 2021.

UK Space Agency, n.d. UK Space Agency International Partnership Programme. Accessed February 24, 2021.

- UK Space Agency, n.d. UK Space Agency International Partnerships Programme Space for Forestry in Developing Countries. Accessed February 24, 2021.
- UK Space Agency, 2020. UKSA IPP ToC. Accessed February 5, 2021.



Itad is a global organisation. Our strategy, monitoring, evaluation and learning services work to make international development more effective. We generate evidence on important issues – from malnutrition to migration – to support our partners to make informed decisions and improve lives.

Itad.com



mail@itad.com

Itad Ltd

Preece House, Davigdor Road, Hove, East Sussex, BN3 1RE, UK +44(0)1273 765 250

Itad Inc

1110 Vermont Ave NW, Suite 500, Washington, D.C 20005, USA

Tel: +1 (301) 814-1492